

Earth's Future

RESEARCH ARTICLE

10.1029/2024EF004902

W. Yao and T. M. Morganti contributed equally to this work and share the first authorship.

Key Points:

- The site-specific context of carbon dioxide removal options is crucial for serious considerations regarding their possible implementation
- Marine carbon dioxide removal options in Germany have the potential to help counterbalance projected future residual emissions
- Further site-specific studies are needed to assess the socio-economic, legal, political and ethical aspects of such implementations

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

W. Yao, T. M. Morganti and N. Mengis, wyao@geomar.de; teresa.morganti@io-warnemuende.de; nmengis@geomar.de

Citation:

Yao, W., Morganti, T. M., Wu, J., Borchers, M., Anschütz, A., Bednarz, L.-K., et al. (2025). Exploring site-specific carbon dioxide removal options with storage or sequestration in the marine environment – the 10 Mt CO₂ yr⁻¹ removal challenge for Germany. *Earth's Future*, *13*, e2024EF004902. https://doi.org/10.1029/ 2024EF004902

Received 21 MAY 2024 Accepted 12 FEB 2025

Author Contributions:

Conceptualization: W. Yao, T. M. Morganti, N. Mengis Data curation: W. Yao, T. M. Morganti, N. Mengis Funding acquisition: W. Yao, T. M. Morganti, A. S. Chua, J. Hauck, F. Havermann, K. Bischof, M. Boersma, U. Daewel, M. Fernández-Méndez, D. P. Keller, A. Kopf, N. Moosdorf,

© 2025. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Exploring Site-Specific Carbon Dioxide Removal Options With Storage or Sequestration in the Marine Environment – The 10 Mt CO₂ yr⁻¹ Removal Challenge for Germany

W. Yao¹, T. M. Morganti², J. Wu³, M. Borchers⁴, A. Anschütz², L.-K. Bednarz⁵, K. A. Bhaumik³, M. Böttcher^{6,7}, K. Burkhard⁸, T. Cabus⁹, A. S. Chua¹⁰, I. Diercks¹, M. Esposito¹, M. Fink¹¹, M. Fouqueray¹², F. Gasanzade⁹, S. Geilert^{1,13}, J. Hauck³, F. Havermann¹⁴, I. Hellige^{15,16}, S. Hoog¹⁷, M. Jürchott¹, H. T. Kalapurakkal¹, J. Kemper¹⁸, I. Kremin¹, I. Lange¹⁶, J. M. Lencina-Avila², M. Liadova¹, F. Liu¹⁹, S. Mathesius²⁰, N. Mehendale¹, T. Nagwekar³, M. Philippi³, G. L. N. Luz¹¹, M. Ramasamy¹², F. Stahl¹⁶, L. Tank⁹, M.-E. Vorrath¹¹, L. Westmark¹¹, H.-W. Wey¹, R. Wollnik²¹, M. Wölfelschneider¹², W. Bach¹⁶, K. Bischof¹⁶, M. Boersma^{3,16}, U. Daewel¹⁹, M. Fernández-Méndez³, J. K. Geuer¹⁵, D. P. Keller¹, A. Kopf¹⁶, C. Merk⁵, N. Moosdorf^{9,12}, N. Oppelt⁹, A. Oschlies¹, J. Pongratz^{14,22}, A. Proelss¹¹, G. J. Rehder², L. Rüpke¹, N. Szarka²¹, D. Thraen⁴, K. Wallmann¹, and N. Mengis¹

¹GEOMAR, Helmholtz Centre for Ocean Research, Kiel, Germany, ²Leibniz Institute for Baltic Sea Research (IOW), Rostock, Germany, ³Alfred Wegener Institute Helmholtz Centre for Polar and Marine Research, Bremerhaven, Germany, ⁴Department of Bioenergy, Helmholtz Centre for Environmental Research, Leipzig, Germany, ⁵Kiel Institute for the World Economy, Kiel, Germany, ⁶SWP-German Institute for International and Security Affairs, Berlin, Germany, ⁷Copernicus Institute of Sustainable Development, Utrecht University, Utrecht, The Netherlands, ⁸Leibniz University Hannover, Hannover, Germany, ⁹Kiel University, Kiel, Germany, ¹⁰Dalhousie University, Halifax, NS, Canada, ¹¹University of Hamburg, Hamburg, Germany, ¹²Leibniz Centre for Tropical Marine Research (ZMT), Bremen, Germany, ¹³Department of Earth Sciences, Utrecht University, Utrecht, The Netherlands, ¹⁴Ludwig-Maximilians-Universität München, Munich, Germany, ¹⁵Max Planck Institute for Marine Microbiology, Bremen, Germany, ¹⁶University of Bremen, Bremen, Germany, ¹⁷Fichtner GmbH & Co. KG, Hamburg, Germany, ¹⁸University of Applied Sciences Kiel, Kiel, Germany, ¹⁹Helmholtz Centre Hereon, Hamburg, Germany, ²¹German Biomass Research Centre, Leipzig, Germany, ²²Max Planck Institute for Meteorology, Hamburg, Germany

Abstract Marine carbon dioxide removal (mCDR) and geological carbon storage in the marine environment (mCS) promise to help mitigate global climate change alongside drastic emission reductions. However, the implementable potential of mCDR and mCS depends, apart from technology readiness, also on site-specific conditions. In this work, we explore different options for mCDR and mCS, using the German context as a case study. We challenge each option to remove 10 Mt CO₂ yr⁻¹, accounting for 8%-22% of projected hard-to-abate and residual emissions of Germany in 2045. We focus on the environmental, resource, and infrastructure requirements of individual mCDR and mCS options at specific sites, within the German jurisdiction when possible. This serves as an entry point to discuss main uncertainty factors and research needs beyond technology readiness, and, where possible, cost estimates, expected environmental effects, and monitoring approaches. In total, we describe 10 mCDR and mCS options; four aim at enhancing the chemical carbon uptake of the ocean through alkalinity enhancement, four aim at enhancing blue carbon ecosystems' sink capacity, and two employ geological off-shore storage. Our results indicate that five out of 10 options would potentially be implementable within German jurisdiction, and three of them could potentially meet the challenge. Our exercise serves as an example on how the creation of more tangible and site-specific CDR options can provide a basis for the assessment of socio-economic, ethical, political, and legal aspects for such implementations. The approach presented here can easily be applied to other regional or national CDR capacity considerations.

Plain Language Summary There is a growing consensus within the scientific community that carbon dioxide removal (CDR) and carbon storage will play crucial roles in global climate change mitigation efforts. Marine-based CDR (mCDR) or carbon storage (mCS) for climate mitigation have gathered significant attention due to their substantial global potential. While numerous studies have assessed global CDR capacities and associated side effects or co-benefits of individual methods, it is important to recognize that global potential does not necessarily translate into local effectiveness. The implementable potential of mCDR and mCS depends



2324277, 2025, 4. Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024EF004902 by Affred Wegener Institut F. Polar- U. Meeresforschung Awi, Wiley Online Library on [2204/2025]. See the Terms and Condition

N. Oppelt, A. Oschlies, J. Pongratz, A. Proelss, G. J. Rehder, L. Rüpke, N. Szarka, D. Thraen, K. Wallmann, N. Mengis Investigation: W. Yao, T. M. Morganti, J. Wu, A. Anschütz, K. A. Bhaumik, T. Cabus, I. Diercks, M. Esposito, M. Fink, M. Fouqueray, F. Gasanzade, F. Havermann, I. Hellige, S. Hoog, M. Jürchott, H. T. Kalapurakkal, J. Kemper, I. Kremin, I. Lange, J. M. Lencina-Avila, M. Liadova, F. Liu, S. Mathesius, N. Mehendale, T. Nagwekar, M. Philippi, G. L. N. Luz, M. Ramasamy, F. Stahl, L. Tank, L. Westmark, H.-W. Wey, R. Wollnik, M. Wölfelschneider, N. Mengis Methodology: W. Yao, T. M. Morganti, N. Mengis Project administration: W. Yao Supervision: D. P. Keller, A. Oschlies, G. J. Rehder, N. Mengis Validation: W. Yao, T. M. Morganti Visualization: W. Yao, T. M. Morganti, N. Mengis Writing - original draft: W. Yao, T. M. Morganti, J. Wu, A. Anschütz, K. A. Bhaumik, I. Diercks, M. Fink, M. Fouqueray, I. Hellige, M. Jürchott, H. T. Kalapurakkal, J. Kemper, I. Kremin, I. Lange, J. M. Lencina-Avila, M. Liadova, F. Liu, S. Mathesius, T. Nagwekar, M. Philippi, M. Ramasamy, F. Stahl, L. Tank, L. Westmark, H.-W. Wey, R. Wollnik, N. Mengis Writing - review & editing: W. Yao, T. M. Morganti, M. Borchers, L.-K. Bednarz, M. Böttcher, K. Burkhard, T. Cabus, A. S. Chua, S. Geilert, J. Hauck, F. Havermann, M.-E. Vorrath, W. Bach, K. Bischof, M. Boersma, U. Daewel, M. Fernández-Méndez, J. K. Geuer, D. P. Keller, A. Kopf, C. Merk, N. Moosdorf, N. Oppelt, A. Oschlies, J. Pongratz, A. Proelss, G. J. Rehder, L. Rüpke, N. Szarka, D. Thraen,

K. Wallmann, N. Mengis

not only on technological readiness, but also on site-specific conditions. Our study explores the capacity of marine-based methods considering local resource availability, geophysical conditions, infrastructure, and land/ sea-area availability to be developed by Germany. We identified 10 proposed options, with half of them being implementable exclusively within German jurisdictions and 3 capable of achieving the 10 Mt CO_2 annual removal target, significantly contributing to Germany's net-zero goal. This underscores the critical importance of considering site-specific contexts in any discussion of mCDR/mCS implementation. Additionally, our study highlights the potential of mCDR/mCS for Germany and calls for further site-specific studies to assess these options beyond natural science or techno-environmental considerations.

1. Introduction

The Paris Agreement (UNFCCC, 2016) requires achieving a balance between anthropogenic emissions by sources and removals by sinks of greenhouse gases in the second half of this century. Such a net-zero goal first and foremost requires substantial reductions and avoidances in greenhouse gas (GHG) emissions. In addition to drastically reducing current emissions, the implementation of carbon dioxide removal (CDR) approaches will play a role in achieving net-zero by counterbalancing residual emissions (i.e., where emission reduction is technologically and/or financially too challenging; Buylova et al., 2021; Fridahl et al., 2020; Mengis et al., 2022; Oschlies et al., 2017). The sixth assessment report of the Intergovernmental Panel on Climate Change (IPCC, 2023) shows that all 1.5°C scenarios applied CDR to reach their goal. However, the CDR options implemented in the scenarios depend, among others, on assumptions that could reasonably be made about costeffectiveness and storage availability of CO₂ (IPCC, 2022). Until now, most considered CDR methods are landbased, relying primarily on afforestation as well as bioenergy in combination with carbon capture and storage (BECCS). Most of the CDR options mentioned in national long-term low-emission development strategies worldwide rely on the expansion or management of existing natural ecosystem sink capacities (Thoni et al., 2020). Accordingly, a considerable amount of research has been devoted to these CDR options, which has resulted in first considerations of limiting factors or bottlenecks. For instance, large-scale, land-based CDR options relying on photosynthetic carbon capture require substantial land area, compete with other land uses such as food, fiber, and energy production, settlement and infrastructure development, and ecosystem services (Boysen et al., 2017; Fujimori et al., 2022; Williamson, 2016). While it is expected that large-scale implementation of any CDR option faces limitations, risks, and biophysical, technical, political and social challenges (Creutzig et al., 2015; Fuss et al., 2018), assessments of marine CDR (mCDR) and marine carbon storage (mCS) approaches within a portfolio of CDR options to reach net-zero emissions are lacking. This omission of marine approaches is a shortcoming, given the high carbon storage inventory of the marine environment and the large fraction of anthropogenic CO₂ that will finally be stored by the ocean.

Germany has set its goal to become GHG neutral by 2045. Optimistic roadmaps to reach this goal provide estimates of residual emissions ranging from 40 to 60 megatons (Mt) carbon dioxide equivalent (CO_{2eq}) annually (Federal Climate Change Act, 2019; Mengis et al., 2022), which corresponds to 5%–10% of Germany's current GHG emissions. These will need to be counterbalanced by the implementation of CDR methods. Less optimistic annual residual GHG emission estimates are even higher, ranging from 45 to 130 Mt CO_{2eq} (Luderer et al., 2021; Merfort et al., 2023). Initial studies of CDR potential within Germany point to a theoretical CDR potential that could reach this scale by employing terrestrial CDR options (Borchers et al., 2022; Merfort et al., 2023). However, the CDR potential on land might be reduced, if current optimistic assumptions about land, fresh water, and energy requirements, as well as the development of storage capacities, are not met – possibly because their implementations would exceed environmental guardrails (Heck et al., 2018).

Here, we aim to complement such efforts by considering the potential of CDR options with storage or sequestration in the marine environment for Germany. To do so, we envision potential implementations of different mCDR categories: (a) the enhancement of the ocean's chemical carbon uptake through alkalinity addition, (b) the enhancement of the "blue carbon" sink capacity (such as in salt marshes, seaweed, or mangrove ecosystems), and (c) offshore geological CO_2 storage coupled with different carbon capture components. We challenge each mCDR option to reach a 10 Mt CO_2 yr⁻¹ removal capacity, as this is a significant fraction of the residual emissions projected for Germany and allows us to assess side effects and challenges, which are likely to be notable on that scale. In addition, by leveling the playing field for all options, we can compare common attributes on Wiley Online I

abrary for rule

applicable Creative Commons License





Figure 1. Roadmap for exploring site-specific carbon dioxide removal options with storage or sequestration in the marine environment pursued in this work.

of the aforementioned mCDR options at the same scale of operation. Our exercise does not aim to provide a comprehensive assessment of mCDR options or a dedicated life-cycle assessment, but rather to present a collection of options that can serve as a basis for further discussion, by making the scale of CDR implementation —relevant to national climate targets—more tangible. By doing so we aim to provide a basis for context-specific discussion and promote further studies that adopt a similar approach in other contexts worldwide.

To this end, we provide first insights into mCDR options on a larger scale, their technological feasibility, infrastructure and resource demands, on the example scale under the site-specific constraints. We first describe how we selected the mCDR and mCS options (Section 2). We then provide a short description of each option, including technological readiness, resource requirements, uncertainties and, if available, cost estimates (Section 3). This is followed by a comparison of the options, expected possible environmental effects, and a discussion of the limitations of our approach (Section 4). For the case study, we conclude that mCDR options are a valuable addition to the net-zero tool box, and that further research is needed to help some of the technology to reach maturity and to reduce the uncertainty in efficiency and environmental effects. Ultimately, the approach we present here may serve as a template for exploring mCDR and mCS feasibility in further contexts worldwide.

2. Materials and Methods

This study aimed to develop site-specific CDR options through three phases: challenge setup, workshops, and synthesis (Figure 1). In the challenge setup, we identified the region of interest (Germany), compiled an initial list of CDR and carbon capture and storage (CCS) methodologies leveraging the knowledge in our research networks (CDRmare and CDRterra), and invited workshop participants. Background research included regional emissions, land/sea area, low-emission energy, material availability, and infrastructure capacity—factors influencing potential deployment (details are given in Section 2.2). These inputs informed our CDR scaling target and assumptions, defining criteria for significant contributions to reaching net-zero in our region of interest. We also designed a unified CDR options template (Table 1) to be employed during the workshop.

The second phase involved two workshops: the first focused on the creation and scaling of the options, generating over 13 options. The latter were screened resulting in merging similar options or excluding options due to



Table 1

Description of the Categories and Parameters Used in the Fact Sheet for the Generation of mCDR Options With the Aim to Reach 10 Mt $CO_2 Yr^{-1}$ Removal; for Details See Supporting Information (SI)

Category	Parameter	Description
Option description	Maturity level	Extent to which an option is available for implementation, following the Technology Readiness Level (TRL) scale (European Commission, 2014) (see Supporting Information \$1 Table \$1)
	Infrastructure	Necessary infrastructure along the chain of implementation
	Biophysical conditions	Necessary environmental conditions for the functioning of the option
	Location	Description of the possible locations and explanation for specific choice made, including the location for resource extraction, material processing, logistic centers, and carbon reservoir
Demand/Input	Area/land	Necessary amount of area on land or ocean
	Material/resources	Necessary amount and type of matter (e.g., rock, soil, etc.)
	Energy demand	Energy demand along the chain of implementation
	Water demand	Necessary amount and type of water
Output	CO ₂ removal potential	If the option cannot reach 10 Mt $\rm CO_2~yr^{-1}$ removal, estimate the maximum yearly $\rm CO_2$ removal rate
	By-products	Additional products with or without market value generated
	Energy output	Energy provision in the form of usable electricity/heat
Environmental impacts	Soils/sediment	Effect of the option causing changes in the state of soils/sediments (e.g., through substance release)
	Water	Effect of the option causing changes in the state of groundwater, runoff water and seawater (e.g., through substance release)
	Air	Effect of the option on the atmosphere (e.g., through release of non-CO ₂ GHGs)
	Noise	Effect of the mCDR option on the ambient noise level
	Ecosystem	Effect of the mCDR option on biota
Cost parameters	CO ₂ removal costs	Marginal removal cost (assuming a fully established system)
	Investment intensity	Investment cost to build at least one unit
	Maintenance cost	Cost for maintenance, including human resources
	MRV costs	Effort and costs for carbon accounting and evaluation of removal
Systemic parameters	max. CO ₂ removal potential	Maximum removal capacity scaling as permitted by constraints (e.g., area, energy, resource limitations)
	Permanence	Carbon reservoir for storage/sequestration (geological, marine biomass, marine soils/ sediments), and the expected length of storage
	MRV capability	The concept MRV is about "Monitoring, Reporting, and Verification." Here, we use this term to describe the ability to measure/estimate carbon fluxes/stock changes and to monitor the environmental impacts, and to verify the amount of removed CO_2

insufficient data. As a result, we generated a list of 10 CDR options, five of which could be implemented in the jurisdiction of the region of interest. The second workshop emphasized detailed option descriptions, discussed approach limitations (session 4.4), and planned the synthesis phase, including metrics for comparison and result formatting.

In the synthesis phase, we conducted comparative analyses (session 4.1–4.3) and an internal review within the research networks.

2.1. CDR Options Development

Following the methodological approach presented by Borchers et al. (2022) with experts from the CDRmare (cdrmare.de) research program, we developed a collection of possible mCDR options for implementation in Germany (Figure 2A1). These mCDR options were generated based on three CDR methods researched in the program: ocean alkalinity enhancement (OAE), blue carbon enhancement (blueCDR), and off-shore geological

LICENS



Earth's Future



Figure 2. Schematic overview of the 10 mCDR options (clustered in three categories).

- 1. Map of Germany indicating the locations chosen for the implementation linked to the region where resources would be available;
- 2. World map for chosen implementation locations for international options (3.1.4, 3.2.2-3.2.4, 3.3.2).
- 3. (B1-3) Schematic drawings of the mCDR options showing sources for CO₂ capture linked to CO₂ storage for the different mCDR categories: ocean alkalinity enhancement (OAE, B1), ecosystem-based carbon dioxide removal in the marine environment (blueCDR, B2), and options with marine geological carbon storage (mCS, B3). The colored dots in B1-3 denote processes in the whole chain of operation, for example, material extraction, material processing, and implementation, while the arrows denote the movement of material down the chain. Note, for option coastal enhanced weathering, the deployment area is both in the North Sea and the Baltic Sea.

carbon storage (mCS) captured from the atmosphere, for example, through bioenergy plants or direct air capture, in collaboration with the research mission CDR terra (cdrterra.de).

The aim was to identify site-specific mCDR options, if possible, within the German jurisdiction, yet, due to option-specific geophysical constraints, some mCDR options are partially (e.g., basalt CO_2 trapping) or entirely (e.g., mangrove planting) located outside of it. Such options would require international cooperation as outlined in Article six of the Paris Agreement (UNFCCC, 2015).

This study describes necessary conditions for each option's implementation, including environmental conditions, infrastructure, technology and resource availability, as well as possible environmental effects. The options were created using a tabular fact sheet, adapted from Borchers et al. (2022), with a unified set of categories and parameters (see Table 1).

At this point, the implementation of OAE, blueCDR and mCS options were envisioned disregarding economic (beyond cost), legal, societal, or political constraints (similar to Borchers et al., 2022), and assuming—where no other information was available—linear scalability to 10 Mt CO₂ removal potentials. The locations in Germany for resource extraction, material processing, and establishment of logistics centers were chosen based solely on



resource and infrastructure availability, which were only used to provide a basis for estimates such as transportation requirements, energy demands, and operational costs. This approach allowed us to fully explore technoenvironmental feasibility and operational locations, necessary infrastructure, and resources under idealized, hypothetical conditions. Although we outline and discuss limitations arising from these assumptions in the overarching discussion, we recommend a detailed assessment of these disregarded factors as the subject of a follow-up study.

2.2. Data Collection, Calculations and Quality Check

Data on mCDR options were assembled by researchers from the German BMBF funded research programs CDRmare and CDRterra, gathered during two workshops (Drübberholz, November 2022; Kiel, June 2023; Figure 1). This expert-driven process brought together participants from various academic disciplines involved in CDR-related research addressing technological, environmental, social, legal, and economic aspects. The participants developed mCDR options based on peer-reviewed literature, reports, resources from their current projects and German-specific information provided by state agencies like the Federal Environment Agency (UBA) and Federal Institute for Geosciences and Natural Resources (BGR). Additionally, literature databases specifically focused on CDR such as the carbon dioxide removal portal (https://carbondioxide-removal.eu/en/ news/) were used. Nevertheless, the list of options based on the three main mCDR methods is not meant to be exhaustive.

The data was compiled in a fact sheet for each mCDR option (see Supporting Information S1). Each option targets an annual removal of 10 Mt CO₂, with all demand, input, and output estimates calculated at this scale. The removal estimates and corresponding demands are derived from existing scientific literature, including laboratory observations, modeling works, method-based synthesis reports, and detailed life cycle analyses. If an option did not reach this scale due to, for example, resource restrictions, a maximum removal estimate (in units of Mt of CO₂ yr^{-1}) was provided and all other estimates related to the demand/input and output categories were calculated accordingly; however, if the option reached 10 Mt CO₂ yr⁻¹ removal, we asked for an estimate of the maximum annual CO₂ removal rate that could be achieved. The cost and technical feasibility were estimated by using the sources described above and supported by our technology consulting partner (fichtner.de). The technology readiness level (TRL) (Supporting Information S1 Table S1), assigned to each method is the lowest level depicted in the whole chain of technologies necessary for the respective mCDR implementation. The permanence, including the form of carbon sequestration, and expected length of storage was also detailed for each option, considering that permanent carbon storage after at least 1,000 years of storage (Brunner et al., 2024). Together with the fact sheets, schematic drawings of the options was developed (see Figures S1-S4 in Supporting Information S1) together with a summary figure (Figure 2). Finally, the experts were asked to provide possible bottlenecks with regard to, for example, resource constraints, and to identify unknowns and research gaps.

3. Collection of Marine Carbon Dioxide Removal (mCDR) Options

The mCDR collection includes 10 options (displayed in 3 classes in Figure 2) for carbon storage and sequestration in the marine environment. In the following, we will briefly describe the functioning of the option, the estimated technology readiness, the resources that would be required to scale this option up to 10 Mt CO_2 yr⁻¹ removal rate, the expected costs for such operations, and key uncertainties (for more details, please see Supporting Information S1). Discussions of the expected environmental effects, and challenges with regard to monitoring, reporting and verification for the options will be in Section 4.

3.1. Ocean Alkalinity Enhancement – mCDR Options Increasing Physical-Chemical Carbon Uptake Through the Addition of Alkalinity

Solutes of weathered silicate and carbonate rocks naturally add alkalinity to the ocean, which increases its CO_2 storage as dissolved inorganic carbon (DIC) (Archer, 2005; Bach et al., 2019). The alkalinity-enhanced seawater at the ocean surface reacts with the CO_2 at the air-sea interface and forms bicarbonate ions, which can be kept in solution for time scales of 100,000 years (Falkowski et al., 2000; Ilyina et al., 2013; Köhler, 2020; Köhler et al., 2013; Renforth & Henderson, 2017). The weathering of alkaline minerals is a negative feedback of the Earth's system regulating atmospheric CO_2 (Archer, 2005; Berner et al., 1983). Ocean Alkalinity Enhancement (OAE) refers to anthropogenic activities with the aim of mimicking the natural process of weathering by adding

23284277, 2025, 4, Downloaded

).1029/2024EF004902 by Alfred Wegener

chung Awi,

Wiley

Online Library on [22/04

(2025].

alkalinity to the ocean to increase its CO_2 uptake (Hartmann et al., 2023; Köhler et al., 2013; Rau et al., 2013) while stabilizing the pH (Hauck et al., 2016; Hinrichs et al., 2023), which has a positive effect on pH-sensitive ecosystems (Albright et al., 2016; Weatherley, 1988).

In the following we describe four OAE options (Figure 2B1): option 3.1.1 explores the possibility of enhancing the alkalinity of the German EEZ (North Sea) through electrolysis of seawater in the presence of silicate minerals, option 3.1.2, explores the possibility of ocean liming in the German EEZ (North Sea), option 3.1.3 explores dissolving basalt powder along the German coastline. Finally, option 3.1.4 explores upgrading existing desalination plants to produce alkalinity. Because this option would only be possible on a small scale in Germany, namely on Heligoland (Germany), we decided to explore this concept in an area outside Germany, where freshwater production from desalination plants reaches a scale of 0.6 km³ yr⁻¹, about the rate related to a CO_2 removal of 10 Mt yr⁻¹, to see its full potential and impacts.

3.1.1. Electrolytic Production and Addition of Alkalinity-Enhanced Solution From Silicate Rock on the German North Sea Coast

Mined and ground silicate rock from quarries in central-Germany could be transported by carrier/barge to an electrolysis facility on the coast in Northern Germany. The North Sea wind farms would be able to provide offpeak renewable energy that could be used for the electrolysis of the seawater with anodes encased by rock powder (details see Rau et al., 2013). The reaction produces hydrogen, oxygen, silicate and magnesium/other-metal salt (solid), and alkaline enhanced (sodium hydroxide, NaOH) seawater (see Supporting Information S1, Eq. 3.1.1a; Rau et al., 2013). The alkalinity-enhanced seawater could then be released to the North Sea. Since the water with enhanced alkalinity needs to be in contact with the atmosphere for efficient CO_2 uptake (Jones et al., 2014), the oceanographic conditions of the German North Sea characterized by shallow depth and high mixing rates (Sündermann & Pohlmann, 2011) would be very amenable to this approach. However, the actual efficiency of CO_2 uptake per unit of alkalinity added in the North Sea would need to be further investigated.

While the technology for electrolysis with brine is already commercially used to produce hydrogen, oxygen, chlorine, hydroxide, and acid for various industries (Lakshmanan & Murugesan, 2014), electrolysis with seawater is currently under development (e.g., Ebb Carbon, 2024; Rau et al., 2018), hence we estimate the TRL of this option to be 3 (according to Table S1 in Supporting Information S1).

To scale this mCDR option to 10 Mt CO_2 yr⁻¹, one would need to dedicate around 30 Mt basalt yr⁻¹ (equivalent to 94% of the current German basalt mining capacity) and around 19 TWh electricity per year (8% of the German renewable electricity production capacity in 2021; House et al., 2007; Rau et al., 2013; for calculations see Supporting Information S1 3.1.1). The mining of the rock would use 0.2 Mm³ of fresh water per year (0.004% of the total groundwater abstraction in 2019; Gerbens-Leenes et al., 2018; Wayman et al., 2021) and a minimum of 460 Mm³ of seawater for electrolysis (Dormann, 2023; 0.001% of the North Sea volume; for calculations see Supporting Information S1 3.1.1). At the same time, the byproducts, namely hydrogen, chlorine, and oxygen, could be utilized and reduce the energy demand and the cost of the option (Rau et al., 2013).

There is no commercial pilot study available for such an option, hence the cost of such an endeavor is highly uncertain. However, in an optimum case scenario, we estimated a cost of 770–1,100 million \notin yr⁻¹ for a 10 Mt CO₂ yr⁻¹ removal rate including mining, transportation, energy, and investment costs without monitoring (for calculations see Supporting Information S1 3.1.1).

One of the remaining challenges regarding electrolysis with seawater is that the constraints on the discharging rate of the alkalinity-enhanced water are unknown. Recent studies recommend values of between 250 and 600 μ mol L⁻¹ to avoid triggering the precipitation of aragonite in coastal regions (Hartmann et al., 2023; Moras et al., 2022). With an assumed 0.86 years of water turnover rate in the North Sea (Sündermann & Pohlmann, 2011), this option would require about 55% of the German EEZ in the North Sea for alkalinity dispersal under a conservative estimation (for calculation see Supporting Information S1 3.1.1) and additional in situ modeling studies may help to reduce the uncertainty. Also, the role of particles as a trigger for precipitation is not clarified yet (Hartmann et al., 2023; Wurgaft et al., 2021). Another challenge is the erosion of the anode and the formation of precipitates on the electrode surface, which reduces the process efficiency (Jamesh & Harb, 2021). Although in the presence of silicate minerals, the occurrence of chlorine and hydrogen chloride would be suppressed, the question of how to safely dispose or use possible byproducts (e.g., up to 7 Mt of chlorine without applying minerals, similar to 3.1.4)

also remains a challenge. Co-emissions from the energy-intensive mining (Moosdorf et al., 2014) could be reduced if the national energy mix is further decarbonized and transportation requirements are reduced if quarries are located close to rivers. Furthermore, a suitable framework for monitoring, reporting and verification (MRV) of alkalinity enhancement is yet to be developed.

3.1.2. Production and Spread of Slaked Lime Along Ship Tracks in the North Sea

In this option, limestone would be mined in quarries in Germany, which are widely spread over the countries adjacent to the North Sea and Baltic Sea. The production of slaked lime $(Ca(OH)_2)$ would then be conducted in the lime quarries on-site. The slaked lime would be transported to a harbor (in northern Germany), where it would be fully mixed with freshwater to produce lime milk. The lime milk is subsequently loaded onto bulk carrier ships and spread into the North Sea along shipping routes. The method could be applied in the German EEZ, and can be extended to the entire North Sea over existing vessel routes.

The production of lime is a mature industrial process with a TRL 9 (Foteinis et al., 2022). The dispersion of alkalinity via ship tracks is less developed. It has been estimated so far mainly based on theoretical discussions with the TRL of 3–4 (Caserini et al., 2022; McLaren, 2012). So currently, we estimate the TRL of this option to be 3.

To reach the 10 Mt CO₂ yr⁻¹ removal target, this option would require a minimum of 17.9 Mt limestone (equivalent to 32% of the current German annual limestone production), 15.27 TWh of electricity (~6% of German renewable electricity/energy production in 2021), as well as 6.78 Mt of freshwater for the slaked lime production per year. The required land for the mining operation is estimated to be 1,858 m² while the sea area needed for the alkalinity spreading is 15,500 km². Under the assumption that mCDR could have a higher priority for the use of these resources, all the requirements could be fulfilled. During the energy-intensive lime production process (with a kiln type that consumes fossil fuels), CO₂ emissions would arise from the calcination process and consumption of fossil fuels (Dowling et al., 2015), which would need to be captured and stored in geological sites (see Section 3.3). Applying less carbon intensive technology (kiln type), and transitioning to renewable energy (Foteinis et al., 2022) would be highly beneficial for this option's efficiency. Low-grade heat is generated during CaO hydration. Currently, its recovery is not practical with the existing lime kilns, but it might be possible with new lime plants (EuLA, 2014).

The total cost for 10 Mt CO₂ yr⁻¹ removal is estimated at 800–1,450 million \notin with CCS applied for the lime production (calculation see Supporting Information S1 3.1.2).

One of the remaining challenges is the decarbonization of the slaked lime production. Even though some processes aiming at zero emission have been proposed (Caserini et al., 2019; Renforth & Henderson, 2017), they still need further evaluation. Another challenge is the assessment of the localized impact produced by slaked lime at the discharge point, due to a temporary increase of pH and alkalinity, of which the environmental side effect is not yet well evaluated and requires additional dedicated experimental studies (Locke et al., 2009; Pedersen & Hansen, 2003). Furthermore, there are presently almost no MRV standards for this practice and a suitable framework thus needs to be built.

3.1.3. Coastal Enhanced Weathering (CEW) Along the German Coast

Similar to option 3.1.1, mined and ground alkaline volcanic rock (e.g., basalt) from quarries in central-Germany (Amann et al., 2022) would be transported by carrier/barge to the coast of northern Germany. For this option, the ground rock powder (less than 10 μ m grain size) would then be deposited on the coastline in a high-energy environment (we take a 100-m-wide band, with an average depth less than 10 m along the beach; see Supporting Information S1 3.1.3). The high energy coastal environment could then weather down the grains of alkaline rocks and thus further reduce the energy requirement and cost of the process (Eisaman et al., 2023; Flipkens et al., 2023; Meysman & Montserrat, 2017). The theory would be that the on-site ground rock will dissolve in the coastal seawater, which enhances the alkalinity and the CO₂ uptake of seawater (Flipkens et al., 2023).

The technological feasibility for conducting CEW is high, since this technique has been applied commercially (TRL 8; "Vesta," 2022). However, the experiment aimed at demonstrating the efficiency of this method has yet to yield accessible results, so as of our study, the concept remains partially unproven. Hence by the time of our study, not all of the concepts are proven, and we assess that CEW has a TRL of 5.

Germany has coastlines extending over 3,700 km, with approximately 3,300 km shallow and suitable for the coastal enhanced weathering approach (Sterr, 2008), which in an optimal condition can sustain the dissolution of 3 Mt of fine basalt powder annually (for calculation see Supporting Information S1 3.1.3). This amount of basalt is roughly 10% of the current German basalt production (BGR, 2021). The total energy consumption for this option is around 614 GWh annually (0.3% of the German renewable electricity production capacity in 2021), which includes mining and grinding, without transportation. The overall CO₂ removal potential is 1.1 Mt of CO₂ yr⁻¹ (for calculation see Supporting Information S1 3.3.3), with the possibility of expansion into the continental shelf (e.g., Fuhr et al., 2024; Hylén et al., 2023).

The cost of 1.1 Mt CO₂ yr⁻¹ removal is around 160 million \notin (excluding MRV cost, for calculation see Supporting Information S1 3.1.3).

Currently, there are no large-scale implementations of CEW that have achieved the 10 Mt CO_2 yr⁻¹ removal target. The verification of the effectiveness as well as the environmental side effects are currently topics of research. The effectiveness would be influenced by the reactivity of minerals (e.g., basalt) under in situ conditions (Ramasamy et al., 2024), the relocation or redistribution of the rock grains (e.g., by wave and tides; Flipkens et al., 2023; Meysman & Montserrat, 2017), or the possibility of spontaneous precipitation of CaCO₃ in the basalt minerals pore waters in lower mixing regimes. Concerning verification, remaining challenges are the scale of deployment with the area coverage and the detection of downstream effects due to the continuous dissolution in seawater. The corresponding research questions include: (a) What are the basalt dissolution kinetics in German coastal habitats under different biological, chemical, and physical variables, and what are the corresponding carbon sequestration timescales? (b) What effects do the release and dispersion of dissolution products have on North Sea/Baltic Sea coastal ecosystems?

3.1.4. Direct Electrosynthesis of Sodium Hydroxide in Desalination Units in Upwelling Regions

In many arid regions or on islands, desalination plants are the primary source of freshwater. During the desalination process, seawater is separated into freshwater and brine. The retained salt in the brine can be split into sodium hydroxide (NaOH) and hydrochloric acid (HCl) through electrolysis. Retaining the acidic HCl on land and reintroducing the brine along with the alkaline NaOH to the ocean enhances ocean alkalinity. In upwelling areas this approach of OAE would reduce CO_2 emissions into the atmosphere by CO_2 -rich upwelled water (Ali et al., 2021; Rau et al., 2013), creating net CDR in the region. While this option could potentially be implemented in a desalination plant on Heligoland, where it would not reach a 10 Mt CO_2 removal scale, we decided to explore its potential in upwelling regions (Sea of Oman) where desalination plants produce more than half a gigaton of fresh water annually (DEWA, 2022). This option's implementation could be connected to Germany through capacity building and carbon trading (please refer to Section 4 for further discussion).

As this approach is currently in the lab-experiment phase with a prototype ("Ebb Carbon," 2024) in development, we assess this option with a TRL of 3.

Utilizing all NaCl from seawater to produce NaOH, achieving 10 Mt $\text{CO}_2 \text{ yr}^{-1}$ removal would require a minimum of 0.47 km³ of seawater at ~35 g kg⁻¹ salinity to be processed by a desalination plant yearly (calculation in Supporting Information S1 3.1.4). For example, in 2020, a single large desalination plant in the United Arab Emirates (UAE) produced 0.62 km³ of freshwater (DEWA, 2022), equating to a processing of 0.6–1.2 km³ of seawater per year (assuming a recovery rate between 50% and 90%). Upgrading such existing desalination plants to remove 10 Mt CO₂ would, however, require 42.7 TWh of additional energy per year (Supporting Information S1 3.1.4). This equals ca. eight times the total renewable electricity production in the UAE in 2020 (IEA, 2023). Therefore, the optimistic capacity of this option at this site would be about 1.3 Mt CO₂ yr⁻¹, limited by the currently available renewable energy. Yet, the demand for freshwater production is expected to increase particularly in arid regions (Baggio et al., 2021), which would likely increase the availability of plants for this option. At an efficiency of 70% of the method, the extraction of 10 Mt CO₂ is expected to produce around 0.6 km³ of freshwater.

Making this CDR option energy- and cost-efficient as well as safe faces several challenges. The 8.3 Mt of the byproduct HCl for 10 Mt CO₂ would need safe disposal and storage. In 2022, the global HCl market was estimated to be 15 Mt with an expected growth of 4.2% by 2032 (ChemAnalyst, 2023). If it was possible to have this CDR option produces H₂ and Cl₂ as byproducts instead of HCl (House et al., 2007; Rau et al., 2018), the by-product Cl₂ gas could be further marketed for various industries (e.g., sewage and wastewater treatment and water cooling systems). The current estimated cost of the electrolysis process (main energy demanding factor) ranges from approximately 900 million \notin ("Ebb Carbon," 2024; Rau et al., 2018) to 3,850 million \notin (Rau et al., 2013; Renforth & Henderson, 2017) for a 10 Mt CO₂ removal rate.

It is currently uncertain how OAE in general and this method particularly could affect the environment (see 4.1.3), how MRV could be established (see 4.1.4) and how to safely store the byproduct of HCl if it cannot be sold. It showcases how large-scale CDR can alter the market of substantial markets for by-products, that is, HCl, or Cl₂.

3.2. Blue Carbon Enhancement – mCDR Options Increasing Carbon Capture and Sequestration by Marine Ecosystems

Marine ecosystems are habitats of efficient primary producers, such as phytoplankton and seagrass. Primary producers can assimilate CO_2 from the atmosphere through photosynthesis and store it in their biomass, underlying sediments, and/or ultimately release the carbon in the form of detritus and/or dissolved recalcitrant carbon sinking to the deep ocean. Kelp forests are estimated to sequester globally about 643 Mt CO_2 yr⁻¹ (224–983 Mt CO_2 yr⁻¹), making them a viable marine CDR option (Krause-Jensen & Duarte, 2016). Though the direct CDR capacity might be limited, these options have co-benefits with no/few disadvantages compared to other CDR measures (Gattuso et al., 2018). The permanence of carbon storage in coastal ecosystems is estimated to range from decades to millennia (Duarte et al., 2013; Fourqurean et al., 2012). Organic carbon sunk into the open ocean is estimated to be out of contact with the atmosphere for years to millennia depending on location, storage depth and general ocean circulation and stratification (Siegel et al., 2021). Permanent carbon storage is considered after at least 1,000 years of storage (Brunner et al., 2024). Expanding or managing of such marine ecosystems for enhanced carbon uptake in coastal areas and the open ocean are considered contributions to blue carbon dioxide removal (blueCDR) activities (Mengis et al., 2023).

In the following we describe four blueCDR options (Figure 2B2): option 3.2.1 focuses on the introduction and expansion of kelp forests in Heligoland, as the only site in the German Bight characterized by rocky substrate as prerequisite to kelp forest establishment. This was the only blueCDR option that we were able to explore with the expertise present and at scale within the German jurisdiction. The (re)establishment of seagrass meadows and salt (tidal) marshes as mCDR options might hold blueCDR potential, especially for non-tropical coastal regions like Germany (Borchers et al., 2022; Macreadie et al., 2021; Stevenson et al., 2023). Nevertheless, since blueCDR options are of large interest and likely a low-regret method (Gattuso et al., 2021), and the potential development of carbon trading schemes under Article six of the Paris Agreement could allow for the exploration of CDR outside one's own territory, we decided to explore some of the options outside of Germany to comprehend the scale and efforts associated. For discussions concerning the implications of such activities, please see Section 4. Option 3.2.2 similar to option 3.2.1 explores the possibility of (re)planting and expanding of coastal ecosystems, in this case mangroves at the coast of Indonesia. For the last two options, we move to recently suggested open ocean blueCDR options (Gouvêa et al., 2020; Wu et al., 2023). Option 3.2.3 explores the use of artificial upwelling (AU) systems to fertilize phytoplankton, enhance productivity and subsequent carbon sequestration through enhanced export. Option 3.2.4 employs the same AU systems to fertilize Sargassum farms in the South Atlantic gyre, where the biomass is subsequently harvested and sunk.

When the chosen reference baseline represents a time when the marine ecosystem was established and not degraded, the suggested (re)establishment approach is considered restoration. This restores degraded ecosystems to their baseline state, which avoids GHG emissions, but is not CDR (Mengis et al., 2023). However, if the reference baseline is the present—meaning the (re)established ecosystem is an "addition" carbon sink to the baseline—it qualifies as CDR.

3.2.1. Introduction of Kelp Forests in the Coastal Waters of Heligoland

The expansion of existing and introduction of new kelp forests poses an option to sequester CO_2 from the atmosphere. Currently, the only kelp site in the German Bight is Heligoland, which could be used as a testing site. For any planting measure, young kelp sporophytes need to be produced for seeding. Therefore, local kelp sporophytes (*Laminaria hyperborea*) would be collected when they are fertile and their spores would be released. The spores would be used to produce "green gravel", little stones seeded with young kelp (Fredriksen et al., 2020). After an initial growth period in the lab, the green gravel would be ready to be brought out to the planting site, where they could be directly dropped from a boat. Several environmental parameters, such as suitable rocky substrate, light availability, water temperature, and nutrients, are required for the growth of kelp (Dean & Jacobsen, 1984; Tittley, 1991). While suitable temperature conditions prevail within most parts of the German Bight (Bolton & Lüning, 1982), a necessary rocky substrate for the kelp to attach to is found only around Heligoland. For further expansion, a rocky substrate would be needed to be established at sites with suitable depth ranges.

The green gravel approach is an established method and already in use in other regions (Alsuwaiyan et al., 2022; Fredriksen et al., 2020), therefore, we rate the TRL between 8 and 9. To achieve the target of 10 Mt of $CO_2 \text{ yr}^{-1}$ removal with this option, an area of about 8,000 km², equivalent to about one-tenth of the German Bight's total area, of kelp forest would be needed (for calculation see Supporting Information S1 3.2.1). The potential area for the introduction or expansion of kelp in Heligoland is around 13 km² (calculation see Supporting Information S1 3.2.1). Most of the coastal area of the German Bight has a muddy substrate, hence the potential within the German Bight is not as big as in other coastal areas. In contrast, most of the shores along the European Atlantic coast have a rocky substrate and would therefore be better suited for kelp planting.

To establish 8,000 km² of kelp forest, about 3,000,000 t of green gravel would be needed. With an estimated price of 6.28 \in per m² of newly established or restored kelp forest, the costs to afforest 8,000 km² would accumulate to a total of about 50 billion \notin (Fredriksen et al., 2020), with an unknown cost for additional substrate on top. This would be a one-time investment, with subsequent annual costs focused on monitoring carbon uptake and addressing potential replanting needs to maintain sequestration in case of disturbances. The material and energy needs are limited to the time of collection of fertile kelp, the cultivation phase, and the deployment of the green gravel.

Currently, one of the major questions is the long-term fate of the carbon captured in the kelp, once the plant detaches. For a long-term storage of CO_2 , it would need to drift to the open ocean and sink deep enough to be sequestered in the sediment (Filbee-Dexter, 2020; Krause-Jensen & Duarte, 2016). However, if the kelp gets washed ashore, it would decompose and the CO_2 would be released back into the atmosphere. The washed-up kelp could potentially be used as a co-product like fertilizer or for the production of bioenergy, but only durable sequestration or storage of the carbon would qualify as CDR (Smith et al., 2023). To further scale up this option within the German Bight, the challenge will be the establishment of a rocky substrate in large coastal areas. However, this option could be applied on many rocky shores in other locations.

3.2.2. Mangrove (Re)establishment in Indonesia

Mangrove forests store a considerable amount of carbon in their biomass and sediments (Alongi, 2014), sequestering ca. 15–73 Gt CO₂ yr⁻¹ globally (Donato et al., 2011). They grow in tropical to subtropical areas, making them currently unsuitable for the German coast. It is estimated that 20%–35% of the world's mangrove forests have been lost over the last 50 years, mostly due to anthropogenic activities and extreme climate events (Polidoro et al., 2010). In this option, the potential of mangrove (re)establishment in Indonesia is explored. Indonesia is chosen as a case study since it currently hosts over 20% of the area of the world's mangrove population (Giri et al., 2011). Over the last 30 years, Indonesia has lost about 8,000 km² of mangroves, and estimates of the initial area of mangrove forests in Indonesia range between 42,000 km² and 77,000 km² about 35 years ago (Ilman et al., 2016), while the current area is about 31,900 km² (Alongi et al., 2016). Therefore, Indonesia has great potential for mangrove (re)establishment and consequently, carbon dioxide capture and storage.

Planting mangroves is a process that is well studied and established, we estimate the TRL to be 9. Before replanting, certain abiotic parameters such as physico-chemical characteristics (salinity, pH), hydrodynamics (waves energy, inundation time) and topography (slope) would have to be checked to select species for planting in suitable zones and improve survivability of the seedlings. Propagules or seeds can be gathered and planted directly, or nurseries can be set up. While nurseries would require a larger time and cost investment, they increase seedling survivability (Hsiung et al., 2024). If necessary, hydrologic conditions might need to be restored to address stressors that have previously caused their declining numbers, by for example, setting up breakwaters, or digging channels. In this case, more time and resources would be needed.

While replanting mangroves describes a simple enough approach, scaling the operation to achieve the 10 Mt CO_2 uptake remains a challenge, since it would require the availability of an area of ~9,400 km² (see Supporting

Information S1 3.2.2). However, Sasmito et al. (2023) showed that only about 1,900 km² are currently suitable for mangrove (re)establishment, which would lead to an uptake of 2 Mt CO₂. Planting mangroves on an area of 9,400 km² would require an investment of 3.4 billion \in (Cameron et al., 2019), which includes 1 billion seeds, planting facilities and infrastructure, and maintenance per km² for a 10 Mt CO₂ yr⁻¹ removal scale.

Remaining uncertainties concerning this approach include the exploration of sufficient areas if there would be an intent to scale up this approach, including the corresponding investment costs. Monitoring, reporting and verification of the carbon sequestration, which is currently mainly driven by the efforts of volunteers, would have to be scaled up and operationalized, for which organizational structures would need to be established.

3.2.3. Artificial Upwelling to Enhance Plankton Production in the North Atlantic

Artificial upwelling (AU) is based on the idea to introduce pipes in open ocean oligotrophic waters to pump up nutrient-rich deeper water to the surface ocean and thereby enhance primary production and export production with the aim to generate an additional CO_2 flux from the atmosphere into the surface ocean.

If this idea (for which we currently estimate a TRL of 2) would be further developed, one possible application of the option could be the North Atlantic Ocean, where long wave-energy powered pipes of 1,000 m length would be installed (see Supporting Information S1 3.2.3 for more info). The necessary infrastructure for this option would include facilities to produce durable pipes made out of steel, plastic or other new materials, ships to install and maintain the pipes in the North Atlantic and a large network of remote sensing technologies (e.g., satellite images and ARGO floats) for MRV purposes (e.g., Mengis et al., 2023). The pipe-covered area needed in the North Atlantic to hypothetically reach 10 Mt CO_2 yr⁻¹ removal via 1,000 m long pipes may be calculated from down-scaled global modeling experiments and would reach a size of 682,000–1,706,000 km² (2–5 times the size of Germany; Jürchott et al., 2023).

This would translate into 40,000 individual pipes and an investment of around 2.2 billion \notin (based on the assessment that one individual 500 m pipe would cost approximately \$ 60.000 and would be able to remove 250 t of CO₂ yr⁻¹ without maintenance costs, see Supporting Information S1 3.2.3 for more information). Although shorter pipes are expected to cost less money compared to longer pipes, longer pipes in model experiments have been found to be more effective in removing CO₂ from the atmosphere (Oschlies, Pahlow, et al., 2010; Yool et al., 2009). The calculated pipe-covered area as well as the costs to reach 10 Mt CO₂ yr⁻¹ removal in the North Atlantic are highly uncertain (also given the low TRL), but some expected side effects and the duration time (see discussion below) of the additionally added CO₂ can already be assessed based on modeling studies. The option is expected to require a total energy demand of 1.4 TWh yr⁻¹ including the pipes production and ship operation (see Supporting Information S1 3.2.3 for more details).

One concern to this AU option, is the fact that the approach would also transport heat and salinity and thereby change the ocean stratification and, eventually, the ocean circulation with potentially substantial impacts on climate (Kwiatkowski et al., 2015). As long as AU is continuously applied, the duration time of additionally stored CO_2 in the ocean is expected to range from decades to millennia (Siegel et al., 2021). It is, however, worth noting that AU, once deployed, would need to be deployed continuously to further increase and keep the additionally added CO_2 stored in the ocean (Keller et al., 2014; Oschlies, Pahlow, et al., 2010). If AU is abruptly discontinued, the surface ocean would immediately respond with CO_2 outgassing, while at the same time additionally stored heat in the ocean interior would radiate back to the atmosphere and within years to decades atmospheric temperatures would rise even above the reference simulation (Oschlies, Pahlow, et al., 2010). Another considerable uncertainty within the development of this option is the durability and stability of the pipes once deployed. As of today, no existing pipe can operate at a depth of 500 m. Another challenge is the assessment of the localized impact that the increase of primary productivity induced by the upwelled water may have in the local food chain. So far, the impact has been studied only at the base of the food web (Ortiz et al., 2022; Tames-Espinosa et al., 2020) and additional experimental studies are required.

3.2.4. Sargassum Farming and Sinking in the South Atlantic Gyres

For this CDR option, holopelagic Sargassum (*Sargassum fluitans* and *S. natans*) would be grown offshore in freefloating aquafarms placed in the South Atlantic subtropical gyre. Nutrients for growth would be provided through artificial upwelling of nutrient-rich deep water from 400 m depth, using upwelling pipes based on the Stommel principle for a perpetual salt fountain (Stommel et al., 1956). This type of artificial upwelling would not require external energy, as the nutrient-rich deep cold water would warm as it comes up and parallelly downwelling warm water from the surface. Environmental conditions (temperature, light, salinity) in this region are favorable for growth of pelagic Sargassum (Gouvêa et al., 2020) and make the implementation of the Stommel upwelling pipe system possible (Kemper et al., 2023).

We estimate the TRL of this option as 2, with the main bottleneck being the cost of the development of the prototype pipes (see Section 3.2.3).

To sequester 10 Mt of $CO_2 \text{ yr}^{-1}$, 57.5 Mt of Sargassum biomass would need to be sunk to the deep sea every year and the total energy demand is 1.7 TWh yr⁻¹ (for calculation see Supporting Information S1 3.2.4). Sargassum increases its biomass by fragmentation at a rate of 5%–10% per day, which means to grow and harvest 0.16 Mt of biomass on a daily basis, a Sargassum standing stock of 4.63 Mt biomass would need to be maintained. To sustain growth, 8,578 m³ of deep ocean water from 400 m depth would need to be upwelled per second (~half of the water discharge of the Mississippi river). This farm would take up an area of 1,324 km². The harvested biomass would be mechanically shredded to extract the nutrients that could be fed back to the aquafarms, reducing the abovementioned amount of deep water needed for fertilization ("Sarga Agriscience," 2021). The shredded biomass will then be pressed into bales and released back to the ocean, where it would sink down, since gas vesicles of the Sargassum would be destroyed in the shredding process, which causes the biomass to lose its natural buoyancy (Baker et al., 2018; Johnson & Richardson, 1977). The logistical effort could be reduced by automated onplatform workflows for farming, harvesting and sinking at the same place.

Due to the low TRL, cost estimations for this option are highly uncertain. Costs of investment and maintenance of infrastructure, including workforce and transportation, would amount to around 1,060 \notin per tonne of CO₂ removed. If scaled-up linearly, this would come up to 10.6 billion \notin for the 10 Mt CO₂ yr⁻¹ removal.

Many uncertainties arise with this option, mainly due to the need to establish the infrastructure, as well as the monitoring. Consequences of impacts on ocean physics and circulation via changes in stratification, as well as from the Sargassum farms and the sunken biomass (e.g., on oxygen levels) are not well understood and need to be addressed in future studies.

3.3. Off-Shore Geological Carbon Storage (mCS) – mCDR Options With Technological Carbon Capture and Subsequent Storage in Geological Formations

Carbon storage in geological formations represents a necessary contribution if technological carbon capture approaches want to achieve carbon removal, like bioenergy carbon capture (BECC) and direct air carbon capture (DACC). In the following, we consider maritime or offshore storage of CO_2 (mCS) in combination with BECC or DACC. Note that geological storage of carbon emissions from fossil or mineral processes (here referred to as fossilCCS following Smith et al., 2023) is distinct from these mCDR approaches, since in this case the CO_2 emissions are avoided, rather than atmospheric carbon being removed; fossilCCS does not constitute a CDR method (Smith et al., 2023).

The concept of geological CO_2 storage is based on controlled injection of dissolved or liquefied CO_2 into porous rocks in the subsurface, so in a geological reservoir. Depending upon the properties/type of the target host rock, the CO_2 can be trapped by several mechanisms, which comprise trapping by an impermeable cap rock or sediment cover (structural trapping), capillary forces in pores (residual trapping), dissolution in water (solubility trapping), and mineral carbonation reactions (mineral trapping).

Two options for geological storage are of particular interest and explored here in more detail (Figure 2B3): option 3.3.1 looks into the possibility of marine biomass for bioenergy generation combined with structural trapping of CO_2 in sandstone formations/saline aquifers that exist in the German North Sea. Deep saline aquifers have a high CO_2 storage capacity due to their regionally large extent, but are still mostly unexplored (Bachu, 2015). Estimates for CO_2 storage in the German EEZ suggest total storage capacities of 4–24 Gt CO_2 (Knopf & May 2017). Time scales of geological CO_2 storage in sandstone formations, while dependent on the regional conditions, can mostly be considered long-term, if not permanent, with a projected minimum of 98% of the stored CO_2 remaining in the reservoir for 10,000 years (Alcalde et al., 2018). Another option (3.3.2), albeit outside of Germany, explores the possibility of DACC combined with mineral trapping of CO_2 through injection of CO_2 into porous basaltic rocks. The basalts form the upper part of the oceanic crust, which is why the majority of them are located offshore in the

23284277, 2025, 4, Downloaded

deep sea and outside the German EEZ. Mineral trapping in the upper ocean crust is an interesting option for CO_2 storage, as basaltic rocks in general have the ability to react more rapidly with CO_2 to produce carbonates than sedimentary sandstones. The accelerated mineral trapping mechanism in basaltic reservoirs compared to conventional sandstone formations facilitates long-term storage by immobilizing the CO_2 (Kelemen et al., 2019). Due to the chemical composition of basalts, which is rich in calcium, magnesium, and iron, the injected CO_2 can react with these elements to form carbonate minerals in the pore space of the rock. Furthermore, impermeable sediments that cover most of the oceanic crust are assumed to impede CO_2 -leakage from the reservoir and, thus, contribute to safe storage conditions. According to the calculations of Snæbjörnsdóttir and Gislason (2016), fractured and porous basaltic flanks of mid-ocean ridges bear a storage capacity of >10⁵ Gt CO₂, exceeding the expected CDR needs manifold. This option would take CO_2 captured in Germany and store it in Norwegian waters, which would require refined agreements concerning CO_2 -trade between Germany and Norway, as discussed in Section 4.

3.3.1. Biomass From Macroalgae Farming for Biomethane Production Combined With Carbon Storage in Saline Aquifers in the North Sea

For a marine feedstock for bioenergy combined with carbon capture and storage, macroalgae could be cultivated and harvested. Plantlets would be set into nearshore floating macroalgae farms (Buck & Buchholz, 2004), which would then be transported offshore and moored within the German EEZ (Buck et al., 2018). The macroalgae would be harvested once a year, and transported to biogas processing and upgrading plants, ideally located in northwestern Germany. The bioenergy plants would need to be retrofitted with carbon capture units, from which the CO_2 is then collected and transported to saline aquifers in the North Sea (with a capacity between 4 and 24 Gt CO_2 ; Knopf & May 2017).

At present, given the highly commercialized coastal seaweed cultivation in Asia, and its absence in North Europe, we estimate the TRL of this option to be 6 (see Supporting Information S1 for estimation).

For this option, to reach 10 Mt CO_2 yr⁻¹ removal, assuming an average annual productivity of 20 kg fresh mass per square meter (FMm⁻²; Buck & Buchholz, 2004; Chen et al., 2015; Chung et al., 2013; Fernand et al., 2017; Kim et al., 2017; Roesijadi et al., 2010), one would need a total area of 2,358 km² (approximately 8.3% of the total area of the German EEZ) for macroalgae cultivation to produce a total of 115 million tonnes macroalgae FM per year (see Supporting Information S1 3.3.1 for calculations). A benefit is the production of biomethane as an energy carrier: although energy is required for harvesting, bioenergy plant operation, and CCS, the model plant produces 4.8 million m³ biomethane, as well as additional heat for external use.

Not considering potential revenues from selling the products, the CO_2 removal cost is estimated at 83 \in per tonne of CO_2 , which would amount to 830 million \in for the 10 Mt CO_2 yr⁻¹ removal scale. If this option was scaled up to this order of magnitude, the construction of new, large-scale (megawatt scale) power plants close to the shore would be reasonable, instead of relying on a big number of decentralized, small-scale plants. While this would impact investment costs, it would reduce the efforts of biomass and CO_2 transportation.

One challenge associated with open ocean macroalgae cultivation is the durability and maintenance of the floating farms, which are susceptible to damage under severe weather conditions in the North Sea (Buck & Buchholz, 2004). One uncertainty of this mCDR option concerns the productivity of macroalgae in the German EEZ, as the estimation given here is highly idealized. Macroalgae growth depends on water temperature, nutrient availability, light and ambient ocean currents. These conditions exhibit significant seasonal variations and are also affected by climate change (Buck & Buchholz, 2004; Grabemann & Weisse, 2008). Previous studies have illustrated that cultivated macroalgae in the North Sea showcase resilience to the high energy environment, even amid severe storm conditions. This implies a potential for macroalgal cultivation within such challenging maritime settings (Bartsch et al., 2008; Buck & Buchholz, 2005; Fortes & Lüning, 1980). Finally, the use of more than 10% of the German EEZ area for the macroalgae production is surely challenging, given the strong competition for area usage within the EEZ. Another challenge for the CDR option is the anaerobic digestion process in biogas plants for macroalgae feedstocks due to unwanted impurities, for example, polyphenols, sulfur, sodium chloride, and heavy metals (Murphy et al., 2015). However, several pre-treatment methods have been suggested in the literature to enhance biomethane yields (Chen et al., 2015; Chung et al., 2013; Suutari et al., 2015).

23284277, 2025, 4, Downl-

3.3.2. Off-Shore Carbon Storage via Mineral Trapping in North East Atlantic Basalts Combined With Direct Air Carbon Capture

For this CDR option, we explore the possibility of CO_2 being captured in Germany by DACC facilities fed with energy generated by offshore wind parks in the North Sea. The captured CO_2 could be transported via cargo ships to an offshore injection site. In the North East Atlantic a basalt volume of approximately 90,000 km³ has been identified, of which 30,000 km³ are in Norway at subsurface depths between 1,500–1503,000m (Planke et al., 2021). For this CDR option, we assume this platform to be located at the Vøring plateau at the North Western Norwegian margin, since the basalts of this region are well studied (Planke et al., 2022) and may have a storage potential of several Gigatons of CO_2 (under investigation; Planke et al., 2021; Rosenqvist et al., 2023). Capturing CO_2 from the atmosphere by using DACC facilities is at the demonstration stage at TRL 6 (IEA, 2022). The concept of offshore CO_2 mineralization still needs to be prototyped and applied in the future. Therefore, the TRL for offshore CO_2 storage in basalts is estimated to be 3–4.

If this option was to be scaled to capture and remove 10 Mt CO₂ yr⁻¹, the DACC would need 18 TWh of energy generated by renewable energy sources (Borchers et al., 2022; Heß et al., 2020), that is ~85% of the energy currently transferred to shore by the North Sea wind parks (Tennet, 2023). The area and the amount of material required to build DACC facilities to capture this quantity of CO₂ are highly uncertain and subject to debate (Chatterjee & Huang, 2020; Realmonte et al., 2019, 2020). For CO₂ transport, the captured CO₂ has to be liquefied (~1.51 TWh/10 Mt CO₂, see Text S3.3.2 in Supporting Information S1) and stored in intermediate tanks. Assuming a distance of 1,800 km, three injection wells, and discharge rates between 1,375 and 2,750 m³ hr⁻¹, transporting 10 Mt CO₂ yr⁻¹ requires ~40 to 140 trips with at least three large, four medium, or eight small cargo ships, respectively. On site, floating production and offloading units (FPSO) with risers for each well are needed for continuous CO₂ injection. Regarding the injection of CO₂, two different approaches exist: either CO₂ can be injected as a "pure" phase or it can be mixed with seawater (see Supporting Information S1 3.3.2). In case of "pure" CO₂ injection, the CO₂ can be heated and compressed by seawater or waste heat recovery on board (to 10°C, 60 bar). Then, compression for injection to, for example, 300 bar, requires 0.094 TWh/10 Mt CO₂ (see Supporting Information S1 3.3.2).

Among others, the costs of capturing CO_2 depend on the DACC technology and the energy costs, which makes the calculation of future capture costs challenging and highly uncertain (IEA, 2022). The same applies for the storage technologies, due to low TRLs future cost estimates are not feasible.

In Iceland, ongoing small-scale projects combine DACC (capturing on the kt CO_2 yr⁻¹ scale) and basalt CO_2 storage onshore ("Mammoth," 2022; "Orca," 2021). While in this option DACC is used for the feed of CO_2 , it is also possible to use bioenergy (similar to 3.3.1). Since storing CO_2 via mineral trapping in offshore basalt formations is still below the prototype stage, many research questions remain: Is the injection of supercritical CO_2 or CO_2 dissolved in water the more suitable option for the Norwegian Sea? Is there an active aquifer in the basaltic layers with flow fostering CO_2 distribution? How does clathrate formation affect the injection scenario? How fast do the carbonates precipitate, how does the reaction affect the pore space geometry and hydraulic properties of the host rock? In which sub-bottom depth is the CO_2 injection safest and most efficient? Getting a profound knowledge of these fundamental questions facilitates finding the best location to drill and estimate the amount of drill holes required to trap a certain amount of CO_2 , which in turn affects the costs to realize this mCDR option.

4. Discussion

The collection of mCDR and mCS options in this study covers different approaches for carbon removal technologies with sequestration or storage in the marine environment. We explore four options for ocean alkalinity enhancement (OAE), four options for enhancing the uptake of blue carbon ecosystems (blueCDR), two of which use artificial upwelling systems (AU), and two options involving geological offshore storage (mCS).

Within the generated mCDR option collection, six out of 10 options could potentially reach the scale of 10 Mt $CO_2 \text{ yr}^{-1}$ removal, of these six, four would have the CO_2 captured within German borders, and three options are located entirely within German jurisdiction (see Figure 2). The blueCDR options based on introducing kelp on Heligoland (3.2.1), and mangrove replanting in Indonesia (3.2.2), as well as the OAE options through electrolysis in desalination plants (3.1.2) and coastal enhanced weathering (3.1.3) did not reach 10 Mt $CO_2 \text{ yr}^{-1}$ removal under the assumptions made here.

From a global viewpoint, the assessment of marine-based CDRs by Gattuso et al. (2021) saw alkalinity enhancement as the front runner when it comes to CDR potentials, but our site-specific assessment highlights that different implementations of the same method can result in differences in the regional capacity. Taking the options for OAE as an example, adding an alkalinity-enhanced solution in the North Sea (3.1.1) and liming along ship tracks (3.1.2) could potentially reach the scale of 10 Mt CO_2 yr⁻¹ removal, while neither coastal enhanced weathering (3.1.3) on the same site nor electrosynthesis in desalination units (3.1.4) on a more suitable site could do it. Similarly, while Gattuso et al. (2021) consider blueCDRs to have very low direct CDR potential globally, our assessment shows that while it is possibly true for kelp forest establishment (3.2.1), other blueCDR options, like mangrove (re)establishment (3.2.2) can reach the megaton scale, and both artificial upwelling (3.2.3) and sargassum farming and sinking (3.2.4) can potentially rise to the challenge.

The techno-environmental comparison of these approaches is challenging, since the options are distinct in both their capture mechanisms – ranging from ecosystem sink capacity enhancement, to enhanced chemical weathering, to technological direct air or point source carbon capture – and their storage processes – ranging from marine biomass, to dissolved inorganic carbon in the water column, marine sediments or varying geological formations. As a result, the mCDR options require different types and amounts of resource inputs, rely on different technologies and infrastructure, and have different co-benefits and side effects. By scaling the CDR options to the same annual removal rate (10 Mt CO₂ yr⁻¹), we attempt to allow for some comparability with respect to factors like area or energy demand, environmental effects and MRV challenges and possibilities, which we will explore in the following.

4.1. Comparison of Marine CDR Options - TRL, Energy and Area Demand, and Bottlenecks

The technology readiness level (TRL) assigned to the mCDR options gives an estimate of the current availability of the technological components of the options, and therefore the development time before possible implementation. The generated collection encompasses mCDR options at all TRLs, with low TRL for both AU options (3.2.3, 3.2.4), and reasonably high TRL for the two blueCDR options managing ecosystems (3.2.1, 3.2.2). While high TRL options are in a rather mature state, having most technological components proven and tested in past and ongoing pilot projects (e.g., IEA, 2022; Raw et al., 2023), low TRL options are still in the concepts' development phase and only some theoretical estimations exist in the literature. For those low TRL options, many of the parameters in this study hence are first-order estimates and therefore rely on reasonable assumptions when scaled up to a 10 Mt CO_2 yr⁻¹, which is a clear limitation to our study, and should be considered when interpreting our findings.

Low TRLs certainly can be considered bottlenecks for most of the mCDR options (e.g., 3.1.1, 3.1.2, 3.1.4, 3.2.3, 3.2.4, 3.3.2), except for the ecosystem-based options 3.2.1 and 3.2.2 with TRLs of 8–9 (Figure 3, Table 2). Other bottlenecks are more method-specific (Table 2). For the high energy demand OAE options (3.1.1, 3.1.2, 3.1.4) as well as for the DACCS option (3.3.2), the main limiting factor would be the supply of renewable energy to decarbonize the process chain, whereas the BECCS option (3.3.1) shows a net energy provision in the form of biomethane and heat. For rock-based OAE options, the considerable demand for material would also pose a challenge.

In contrast, the blueCDR options would have low energy demands, since the energy demand in nurseries or upwelling pipes is relatively small, but would require considerable marine space for their implementation. This limitation prevented both ecosystem-based options (3.2.1, 3.2.2) to reach the 10 Mt CO_2 yr⁻¹ removal scale. If marine biomass for BECCS or Sargassum farming are to be employed to relieve pressures on land, the engineering for suitable open ocean farms for seaweed in the North Sea remains a challenge. The area and energy demand of harvesting and transporting 115 Mt of macroalgae biomass could potentially present another bottleneck in an at-scale implementation due to possible conflicts with other sea areas usage (e.g., wind parks). In addition, technological innovation in the open ocean engineering, deployment and maintenance of the farm structures and upwelling pumps would be necessary before these options can be implemented. Similarly, spreading alkaline solutions in the ocean requires access to large areas (3.1.1, 3.1.2, 3.1.4).

In contrast, area demand of the DACCS option is less than 100 km² (3.3.2), with the highest demand arising from the DACC plants. While the area demand is often assumed to be less problematic for mCDR options compared to land-based CDR options, challenges with respect to accessibility of large ocean areas (3.1.1, 3.1.2, 3.1.4, 3.2.3)



Figure 3. Summary and comparison of mCDR options at a scale of 10 tonnes of CO_2 removal. This figure shows net energy demand against area usage (a); and technology readiness level (white circles) against whether or not the option would be implementable in German jurisdiction and could reach a scale of 10 MtCO₂ yr⁻¹ (b). The area usage displayed here includes estimated land and sea surface area. Net energy demand is given as demand minus supply, which can lead to negative values (produced energy). We chose to compare our options here on the basis of 10 tonnes rather than 10 megatonnes, to allow for a comparison at a scale that all the options could reach, and that would be significant to compensate for one person's annual emissions in Germany (8.7 t yr⁻¹ for 2021).

and potential conflicts with other uses (e.g., marine protected areas, offshore wind, etc.) in more coastal areas (3.1.3, 3.2.1, 3.2.2, 3.3.1) remain (Table 2).

For the BECCS and DACCS options (3.3.1, 3.3.2), the lack of existing infrastructure (carbon capture facilities, transport infrastructure and offshore platforms) for the geological carbon storage as well as the high energy demand for the carbon capturing process limit their potential (Table 2).

4.2. Comparison of Marine CDR Options - Environmental Effects

The mCDR options aim to alter biological, chemical or geological processes with the goal to increase carbon uptake, yet alongside this desired effect likely come a variety of unintended environmental effects of varying predictability. The extent of the possible environmental effects at a specific scale at a specific site is difficult to quantify without dedicated studies (e.g., model studies with site-specific boundary conditions) and will have to be subject to further investigation. We indicated the research gaps and uncertainty related to possible environmental effects (Table 2) and we compiled here possible expected environmental effects gathered from the literature.

The possible impacts of OAE on the wider marine ecosystem on short to long time scales require more research (Albright et al., 2016; Bach et al., 2019; Cripps et al., 2013; Ferderer et al., 2022; Gately et al., 2023; NAS, 2021). The introduction of alkaline substances into seawater could allow for additional CO₂ uptake while stabilizing the pH, although this does not reverse previous acidification (Hinrichs et al., 2023; Hutchins et al., 2023). The addition of alkalinity is reported to have a positive effect on ecosystems that are sensitive to ocean acidification (Albright et al., 2016; Weatherley, 1988), however, it is also shown that less dissolved CO_2 may reduce growth rates of calcifying organisms (Langer et al., 2006). At the point of alkalinity injection, OAE might cause localized temporary pH and alkalinity spikes, which might be ecotoxic and detrimental for the affected ecosystems (Locke et al., 2009). On a longer timescale, such an intervention could impact the physiology of marine organisms and the ecosystem structure (Roberts et al., 2010). If OAE options introduced minerals or their solution into the ecosystem (3.1.1, 3.1.3), this would likely have a fertilization effect (Hauck et al., 2016; Köhler et al., 2013). OAE therefore could increase the primary productivity on site, but decrease oxygen level and increase acidification in the water column downstream (Oschlies, Koeve, et al., 2010). Depending on the geochemical composition of the used rock, heavy metals could also get into the water through mineral dissolution or electrolysis and desalination processes (Arribére et al., 2003; Lattemann & Höpner, 2008), which needs to be regulated. In the case of upgrading existing desalination plants, the discharge from plants (3.1.4) might be contaminated with filtercleaning products (anti-scalants and antifoulants) which would impose potential danger to the local environment (e.g., Ahmed & Anwar, 2012; Al-Anzi et al., 2021; Jones et al., 2019). Furthermore, indirect effects from OAE options due to mining activities (3.1.1, 3.1.2, 3.1.3) would likely negatively impact soil, air, and water quality on land, and introduce noise pollution to the environment on and off site (Sengupta, 2021). Also, there are

AGU
ADVANCING EARTH AND SPACE SCIENCES

Overview of Bottlenecks and Re.	search Gap/Unce	ertainty Iden	tified During	the Developme	nt of the Ten mCDI	R Options				
Category		70	AE			blueCDR			m0	S
Option	3.1.1 Alkalinity- enhanced 3 solution al	.1.2 Ca(OH) ₂ long ship tracks	3.1.3 Coastal enhanced weathering	3.1.4 Electrolysis in desalination units	3.2.1 Kelp forest establishment	3.2.2 Mangrove (re) establishment	3.2.3 Artificial upwelling	3.2.4 Sargassum farming and sinking	3.3.1 Macroalgae farming for biomethane	3.3.2 Carbon storage via mineral trapping
Bottleneck factor for reaching 10 Mt CO2 per	year removal capacity									
Infrastructure/Technology/TRL	TRL = 3. T Electrolysis technology for commercial usage.*	RL = 3. Installation of tanks, pumping and piping systems on ships.*	TRL = 5. Transport, grinding and distribution of minerals at coastline.*	TRL = 3. Electrolysis technology for usage.*			TRL = 2. Pipe efficiency under development.*	TRL = 2. Upweling infrastructure under development.*	TRL = 5. Farming infrastructure in the open ocean.*	TRL = 3.4. Infrastructure to capture, transport and inject CO ₂ has to be built.*
Land/sea area demand	Limited area for imple For 3.1.4 there is pos	ementation and it marine pr ssible conflict wit proc	s possible conflict w sected areas)* h space on land (sat luct HCl)	ith other uses (e.g., è storage of waste	Possible conflict with other sea area usage ⁴ Area availability for creation of suitable substrate [*]	Possible conflict with other land or sea area usage*			High area demand and possible conflict with other uses (e.g., marine protected areas)*	Potential high area demand on land for DACC facilities
Material demand	Mining and product	tion capacity coul scaling up*	d restrict further			Large scale supply of propagules and seedlings for implementation				Material needed to build and run DACC facilities to achieve large scale capture
Clean Energy Demand	Decarbonation of the CDR efficiency Emission during mini reduces the CDF	slaked lime prod ng, transportation R efficiency	uction reduces the , and deployment	Availability of renewable energy*				Off-shore renewable energy (e.g., floating solar) still under development		High energy demand for DACC
Research gap/uncertainty										
Environmental effect (±)	Unclear effects of hig (e.g., pH, nutri	her alkalinity and ent level, trace m	temporal increase c etal, and salinity) or	of water parameters 1 marine life	Increases diversity in establ fīsh, invertebrates, birds, n alternative ecosystem.	shed areas. Offering shelter for tammals. Establishment of an Coastal erosion protection	D	nknown	Unkn	омп
CDR effectiveness	Alkalinity injection ra to area demand For 3.1.3 option im effi	ate and dissolution () to prevent secon sequestered c plementation of C ciency without a	a kinetics need furth adary precipitation. arbon is uncertain 2CS for the lime pro low carbon energy :	ter research (linked Long term fate of duction. Method source			Method efficiency needs more research	Fate of sunken biomass in the deep sea is unknown	Method efficiency needs more research	Method efficiency without a low carbon energy source
MRV		Not estab	lished yet		High effort du	e to the vast area	Not es	tablished yet		
Cost	No pilot plant available			Cost of electrolysis technology at commercial level			Upwelling r	sipe cost uncertain		No pilot projects exist, which makes storage cost estimates uncertain
Durability	Long-term stabili alkalini	ty of added ty	Keeping the weathering material within the reactive (wave) zone	Long-term stability of added alkalinity	Long-term faite of the sequestered carbon is uncertain	Long-term stability of stored carbon depends on the system's stability (subject to climatic changes and human activities)		Depends on the pre- treatment of the biomass before sinking and the remineralization rate at the deep sea		Long-term evolution of pore space and permeability
Other	Safe disposal of waste			Safe storage of waste product HCI	Uncertainty in the area estimation	Uncertainty in area estimates (strong local variations)	Lack of per interna	mits to operate in trional waters	Macroalgae productivity in the region	DACC capacity dependency

Earth's Future

Table 2

Note. "*" marks the current factor(s) restricting further upscaling of the option.

23284277, 2025, 4, Downloaded

concerns about health risks associated with finely crushed $(1-10 \ \mu m)$ material containing fibrous serpentine minerals like asbestos, as well as potential problems with wind-borne transport of fine ground olivine (Hangx & Spiers, 2009).

Environmental effects from blueCDR options could include changes in species compositions, light availability for organisms living on the seafloor or ambient nutrient levels. Furthermore, if biomass is mobilized, its decomposition would cause a decrease of oxygen levels in the adjacent deeper water or on the seabed. However, the introduction of kelp or mangrove forests would provide several co-benefits such as coastal protection, provision of services like food, timber or medicine, provision of habitat for (commercially important) fish species, water purification (Castro et al., 2022; Kayalvizhi & Kathiresan, 2019; Theuerkauff et al., 2020), as well as cultural services, including tourism, religion, and contributing to general well-being (Bandaranayake, 1998; Cuba et al., 2022; Eger et al., 2023).

The environmental effect of AU options is highly uncertain and has so far mostly been assessed within modeling studies. Ocean fertilization from AU options, introducing nutrient-rich deep water to the surface, may increase primary production and shift plankton community structure (Baumann et al., 2021; Ortiz et al., 2022). An increase of primary production results in an increase of carbon export and remineralization that could reduce oxygen concentration in deeper waters and increase GHG release (e.g., methane and nitrous oxide; Williamson et al., 2012). Additionally, the artificial upwelling of water may vertically transport microbes, likely impacting the microbiological environment. Changes in temperature, salinity, stratification, and circulation induced by AUs require meticulous study due to their irreversible nature (Oschlies, Pahlow, et al., 2010).

Environmental effects from large-scale macroalgae or sargassum farms are still understudied, but likely include increased or changed biodiversity both at the surface and in the deep sea (Baker et al., 2018; Casazza & Ross, 2008), increased albedo (Bach et al., 2021), reduced light at the surface and possibly co-emission of halocarbons (Keng et al., 2013; Mithoo-Singh et al., 2017). While growing, Sargassum excretes large amounts of dissolved organic matter (DOM) (Powers et al., 2019). Some of the excreted compounds are likely to persist in the ocean as recalcitrant DOM, which would contribute to CDR due to its persistence against microbial degradation (Buck-Wiese et al., 2023). It has also been shown that existing off-shore Sargassum ecosystems can also contribute to increasing biodiversity and providing a habitat for several species such as turtles, dolphins, and several fish species (Martin et al., 2021). A careful evaluation of open ocean macroalgae farms considering impacts and potential co-benefits on marine ecology, biogeochemistry and fishery is required (Chung et al., 2013; Fernand et al., 2017; Gao et al., 2021; Wu et al., 2023). This includes accounting for offsets from the remineralization of particulate organic carbon (POC) export, the generation of halocarbons, calcification by encrusting marine life, and changes in surface albedo (Bach et al., 2021; Chen et al., 2020; Jia et al., 2022; Krause-Jensen & Duarte, 2016; Pedersen et al., 2021; Wada et al., 2015; Wang et al., 2023). Potential oxygen depletion through biomass remineralization and potential methane and hydrogen sulfide production in the deep sea have been proposed (Levin et al., 2023), but are a subject for further studies.

The expected environmental impacts of offshore geological carbon storage are mainly noise (e.g., Marappan et al., 2022) and CO₂ leakage events. Noise is generated by drilling and pumping or may be produced if active seismic methods are used to explore and monitor the storage site. Passive seismic methods have the potential to reduce noise stressors (Goertz-Allmann et al., 2014). Leakage, in the case of dissolved CO₂ injection, is prevented by the higher density of CO₂-charged seawater compared to normal seawater. A higher leakage risk arises if CO₂ is injected as pure phase. Results of a controlled CO₂ release experiment near the Sleipner CO₂ storage site showed that, in case of leakage, CO₂ gas bubbles are dissolved within 2 m above the seafloor and the excess dissolved CO₂ is further dispersed by tidal currents (Vielstädte et al., 2019). Their model indicates that pH changes exceeding seasonal changes are only found within a distance of approx. 80 m from the well. Still, particularly for prolonged leakage and higher CO₂ release rates, increased CO₂ concentrations and low pH bottom waters could have noxious effects on benthic organisms in the vicinity of a leaky well (Vielstädte et al., 2019). The risk of CO₂ leakage is reduced if fast crystallization processes are triggered by the injection.

It is noteworthy that in contrast to land-based CDR options, the environmental impacts of mCDR options are even less constrained by the deployment site due to the continuous ocean medium (Mengis et al., 2023). The blueCDR (including AU) and OAE options, in particular, might not only affect the region of the operation, but could also cause changes downstream as the water masses move (e.g., Berger et al., 2023; Wu et al., 2023), causing challenges for the long-term monitoring and verification of carbon storage (Mengis et al., 2023).

4.3. Comparison of Marine CDR Options – Evaluation and Monitoring of mCDR

The ultimate goal of CDR is to remove CO_2 from the atmosphere. Hence it is important to know how much of the carbon sequestered and stored would translate to reduction of the atmospheric carbon dioxide pool (Bach et al., 2023). The options in this study differ in permanence: while OAE and mCS options can last over 10,000 years, blueCDRs exhibit greater variability, ranging from centuries to millennia (see Tables S1–S12 in Supporting Information S1). Comprehensive evaluation and monitoring would be needed to accurately assess the effectiveness and side effects of the mCDR options. Presently, no standard monitoring protocol for mCDR options is in place.

However, for OAE a best practice guide on responsible research including MRV has been published (Ho et al., 2023; Oschlies et al., 2023). MRV for OAE options would need to consider in situ pre-conditions. At the release site, mooring stations equipped with autonomous systems to monitor the carbonate system and biological components could provide initial alkalinity signals. Existing observational networks like the Ship-of-Opportunity, FerryBox-integrated, membrane-based sensor measurements in the surface North Sea, with the measuring instruments equipped on repeating commercial vessel, could provide a cost-effective way to observe the surface ocean at a relatively large temporal resolution and spatial coverage (Macovei, Petersen, et al., 2021). However, in the 10 Mt CO₂ removal scale, if the added alkalinity would spread in the North Sea evenly, it would be difficult to verify the effect on total alkalinity, since the expected change (4.3 µmol/L) is much smaller than the natural seasonal variation of alkalinity on site (Hoppema, 1990), smaller than the alkalinity sensor accuracy (Sonnichsen et al., 2023) and on par with the current laboratory alkalinity measuring techniques (Bockmon & Dickson, 2015). This means that one would depend on models alongside the observational effort near the discharging site (Ho et al., 2023). Accompanying the evaluation of the mCDR effect, environmental monitoring (e.g., water quality monitoring and fishery management) needs to be in place, due to the various potential side effects of OAE (see Section 4.1.2).

Challenges and ways forward concerning the evaluation and monitoring of blueCDR approaches have recently been outlined (Mengis et al., 2023). To evaluate, for example, the CDR potential of Sargassum aquafarming coupled with AU, the flow of CO_2 from the atmosphere to the Sargassum biomass needs to be demonstrated using surface ocean pCO_2 sensors and flux calculations based on gas-exchange parameterizations, as well as the permanence and stability of the biomass in the deep sea. After establishing key concepts, surface, submerged biomass stocks, biodiversity, bycatch and environmental parameters (nutrients, trace elements, pCO_2 , O_2 , DOM fractions) need to be monitored regularly to spot possible impacts and environmental changes. Environmental parameters can be collected to implement the data in a predictive model. Any MRV for AU would be highly challenging, since the additionally stored CO_2 in the interior ocean will move with the currents and get diluted (Mengis et al., 2023).

Finally, in terms of monitoring geological storage sites of mCDR options, many of the developments in petroleum reservoir monitoring could be adapted. The now widespread use of time-lapse seismic reservoir monitoring (Lumley, 2001), as demonstrated at the Sleipner project (Arts et al., 2004), time-lapse (4D) seismic monitoring, gravity field monitoring, surface gas monitoring, and distributed fiber-optic sensing are just some possibilities. Furthermore, the monitoring of storage sites can profit from existing regional geophysical monitoring, local deployment of landers that are equipped with sensors, for example, DIC sensors, or isotope measurements of cores from monitoring wells to confirm the carbonization reaction.

4.4. Limitation of This Approach – Considering Economics, Ethics, Acceptance, and Legality of mCDR Collection

Our focus lies on questions of effectiveness, scalability, and technological feasibility combined with some information on costs and environmental effects. Yet, there are other important questions that arise about these mCDR options and new technologies and practices. Even though an in-depth assessment is beyond the scope of this study, we want to briefly highlight four aspects we deem to be particularly pertinent for the potential deployment of mCDR or mCS: the economics, ethical arguments for or against deployment, societal acceptability, and legality.

The cost estimates for early-stage deployment or piloting, we can provide here vary widely between the different mCDR options. Since many concepts are not yet implemented at scale, and/or have a rather low TRL, there are

considerable uncertainties associated with these cost estimates. The ranges of the estimates for operative costs are substantial across different studies (NAS, 2021). Furthermore, when evaluating economic aspects of mCDR methods, it is necessary to go beyond operative cost assessments based on current prices and also account for price effects after a large-scale roll-out. For example, an increased demand for certain input materials will increase the price of these materials, thus affect the removal costs and the relative price for the mCDR option compared to other emissions reduction and CDR options and thus, the overall costs for reaching national climate targets (Klepper & Rickels, 2012). Other factors such as learning-by-doing, the permanence of CO₂-storage, and the cost of negative side effects and the value of co-benefits should also be considered (Rickels, 2023). An integrated economic analysis of mCDR deployment scenarios is urgently needed, but beyond the scope of this work.

Ethical convictions, namely on the impermissibility of letting people suffer the full consequences of unabated climate change, are a major part of what motivates interest in CDR. Implementation scenarios of CDR options can be evaluated from an ethical perspective, for example, looking at their impact on people and the natural world as well as their governance (Heyward, 2019). This paper affirms the value of these considerations (Lenzi et al., 2018) but restricts itself to assessing the techno-environmental feasibility and effectiveness of certain CDR options. This is a paper about what could potentially be done – it paves the way for later questions about what should be done (Tank et al., 2025; Zimm et al., 2024). In this context, one important issue that needs to be briefly discussed here, is the question of the moral (im)permissibility of the inclusion of CDR options outside of Germany or its territorial waters. One could argue that CDR options tasked with counterbalancing German emissions should be deployed on German territory, and not 'outsourced' to other countries. We want to highlight that the ethical implications of the extraterritorial use of CDR will heavily depend on the specific characteristics of the projects in question. Disregarding side effects on local people or the local environment abroad because they would happen 'somewhere else', would clearly be morally problematic. However, if the options in question receive the informed consent of the local inhabitants, and especially those potentially affected (Preston, 2013), we deem it an open question whether extraterritorial CDR, potentially in areas where its effectiveness is much higher than in Germany, could be morally permissible. While an exhaustive discussion of this issue is beyond the scope of the paper, we do see the need to discuss this further and hope to provide input for these debates.

Any CDR implementation happens within societal context, which involves opinions of the general public and of local communities affected by the measure (Chen et al., 2015; Segreto et al., 2020). Concerns that remain unaddressed and voices that remain unheard can negatively influence the socio-political feasibility of implementation (Wüstenhagen et al., 2007). Considering local knowledge and contexts such as governance structures, past experiences, and enabling participation through benefit-sharing on the ground can increase the chances for long-term success (Merk et al., 2022). Ideally, this would require time and financial investment prior to starting the project to organize participatory engagement workshops and information sessions to enable affected communities to provide input on project siting and planning processes (Satterfield et al., 2023). Societal engagement in the project planning process needs to be built on trust, which can be gained by adhering to norms of procedural justice like transparency and fair participation (Heyward, 2019). This starts with the transparent communication about moral, social, economic and environmental risks and benefits and the possibility to participate in decisionmaking processes. While our analysis strives to make basic risks and benefits transparent, the development of tools for co-producing knowledge (Norström et al., 2020; Satterfield et al., 2023) and for supporting informed decisions (Nanz & Fritsche, 2012) are beyond the scope of this paper.

Turning to questions of legality, mCDR deployment raises a dilemma. While mCDR poses immediate risks, renouncing to deploy it may leave climate change unabated and create risks in the future. Delayed emission reductions reinforce our dependency on mCDR to reach the 2100 target of 1.5°C of global warming and will exacerbate the sense of urgency in choosing the lesser evil. This ethical dilemma is naturally reflected in the applicable laws. On the one hand, 'traditional' environmental law discourages any activity that may have adverse effects on the environment. On the other hand, climate change law, which sets ambitious temperature goals, arguably supports the enhancement of sinks and reservoirs. The precautionary principle for instance can be interpreted to either prohibit mCDR as a precaution for safeguarding the integrity of marine ecosystems, an understanding that seems to be envisaged by the Convention on Biodiversity. At the same time, it can also be understood as encouraging, or even requiring, the implementation of mCDR options in light of the consequences of unabated climate change as established by Article three of the UNFCCC (Tedsen & Homann, 2013). Contemporary international law therefore needs to reposition itself in order to adapt to this new, more complex reality in which the status quo may no longer be the safest choice. In that regard, the 2009 amendment of the



London Protocol to allow for the geological sequestration of CO_2 and the 2013 amendment on marine geoengineering demonstrate the ability of international law to evolve. At the German level, current legislation, written with strong environmental concerns in mind, act as showstoppers for the deployment, and even research, of mCDR (Ginzky et al., 2011; UBA, 2023). Developments on the matter of mCDR at the international and European levels could thus provide guidance for an innovative interpretation of the precautionary principle, in which research is enabled and risks are controlled through a regime of safeguards.

5. Conclusion and Outlook

Our study represents an example on how to create context-specific CDR options for a more tangible and concrete basis for further studies and discussions. We believe that this format warrants application in many more contexts worldwide to address the global challenge of net-zero at an actionable level of detail. The resulting study is the first attempt at developing site-specific CDR options with storage or sequestration in the marine environment for a German context. We challenged the mCDR options to reach a scale of 10 Mt CO_2 yr⁻¹ removal, which would represent a substantial contribution to Germany's net-zero goal by counterbalancing projected residual emissions. This approach allowed us to compare mCDR options on the same scale and in their actual context of implementation, thereby providing a more policy-relevant evaluation of the associated area, energy, and resource demands.

We find that six out of the 10 options considered in this study could potentially reach an annual removal rate of 10 Mt CO₂. Among them, three appear feasible within Germany: electrolytic production and addition of alkalinity-enhanced solution from silicate rock; production and spread of Ca(OH)₂ along ship tracks in the North Sea, and biomass from macroalgae farming for biomethane production combined with carbon storage in saline aquifers in the North Sea. This study does not consider the GHG emission life-cycle, which is not ideal when it comes to the comparison of CDR effectiveness. However, due to the varying TRLs, many of these options cannot get robust emissions factors for life-cycle exercises, without further research. This study does not exhaust all possible mCDR options in Germany nor assess their theoretical maximum potential for Germany, but with six options passing the 10 Mt CO₂ removal yr⁻¹ benchmark, we could envision that mCDR can provide a significant contribution to carbon removal, and should be considered in the option portfolio for German net-zero.

However, we also identified a multitude of bottlenecks concerning mCDR options for annual 10 Mt CO₂ removal, ranging from geophysical constraints to current material and clean energy availability to the current technology and infrastructure capacity (see Table 2). For example, with the exception of kelp forest management, the CDR potential of these options cannot currently be fully determined due to their low technological maturity, leading to high uncertainty in our estimates. Remaining research questions are method-specific: for OAE options, research concerning the understanding of the dissolution process of implemented materials is needed, which in turn will impact the spreading mechanisms, thereby impacting area, resource, and energy demands. For OAE and blueCDR options, one of the biggest remaining questions concerns the monitoring and verification of carbon fluxes. For the geological storage of carbon in the German EEZ, pilot studies are required to explore potential storage sites and determine achievable removal rates. For all of these options, thorough cost analyses along with GHG footprint life-cycle assessments are necessary to provide a realistic assessment of investment and market costs.

This study intends to provide a collection of mCDR options as the basis for more thorough assessments of mCDR options developed by Germany, both within and outside of German borders. Ideally, these future assessments will be supported by more comprehensive implementation scenarios that include evaluations of social, ethical, and political impacts. In addition, multiple CDR options should be jointly considered for potential synergies and trade-offs, especially regarding additivity and concomitant side effects, to better characterize possible future implementation scenarios.

Our exploration of mCDR options on a 10 Mt CO₂ yr⁻¹ removal scale has revealed a multitude of limitations, bottlenecks, and uncertainties. We believe more of such assessments are needed to bring the expectations about CDR option down-to-earth (Dean et al., 2021), because over-optimistic or untested assumptions about large-scale CDR implementation should not serve as a reason to delay emissions reduction by suggesting it is possible to "emit now and remove later" (Fuss et al., 2014; Williamson, 2016).



Data Availability Statement

This study did not involve the use of any datasets or computational models. The research is based on conceptual analysis and literature synthesis, with all sources explicitly cited in the main text of the article and the supplementary information.

References

- Ahmed, M., & Anwar, R. (2012). An assessment of the environmental impact of brine disposal in marine environment. International Journal of Modern Engineering Research (IJMER), 2(4), 2756–2761.
- Al-Anzi, B. S., Al-Rashidi, A., Abraham, L., Fernandes, J., Al-Sheikh, A., & Alhazza, A. (2021). Brine management from desalination plants for salt production utilizing high current density electrodialysis-evaporator hybrid system: A case study in Kuwait. *Desalination*, 498, 114760. https://doi.org/10.1016/j.desal.2020.114760
- Albright, R., Caldeira, L., Hosfelt, J., Kwiatkowski, L., Maclaren, J. K., Mason, B. M., et al. (2016). Reversal of ocean acidification enhances net coral reef calcification. *Nature*, 531(7594), 362–365. https://doi.org/10.1038/nature17155
- Alcalde, J., Flude, S., Wilkinson, M., Johnson, G., Edlmann, K., Bond, C. E., et al. (2018). Estimating geological CO2 storage security to deliver on climate mitigation. *Nature Communications*, 9(1), 2201. https://doi.org/10.1038/s41467-018-04423-1
- Ali, E. B., Skjelvan, I., Omar, A. M., Olsen, A., de Lange, T. E., Johannessen, T., & Elageed, S. (2021). Sea surface pCO2 variability and air-sea CO2 exchange in the coastal Sudanese Red Sea. *Regional Studies in Marine Science*, 44, 101796. https://doi.org/10.1016/j.rsma.2021.101796 Alongi, D. M. (2014). Carbon cycling and storage in mangrove forests. *Annual Review of Marine Science*, 6(1), 195–219. https://doi.org/10.1146/
- annurev-marine-010213-135020
- Alongi, D. M., Murdiyarso, D., Fourqurean, J. W., Kauffman, J. B., Hutahaean, A., Crooks, S., et al. (2016). Indonesia's blue carbon: A globally significant and vulnerable sink for seagrass and mangrove carbon. Wetlands Ecology and Management, 24(1), 3–13. https://doi.org/10.1007/ s11273-015-9446-y
- Alsuwaiyan, N. A., Filbee-Dexter, K., Vranken, S., Burkholz, C., Cambridge, M., Coleman, M. A., & Wernberg, T. (2022). Green gravel as a vector of dispersal for kelp restoration. *Frontiers in Marine Science*, 9. https://doi.org/10.3389/fmars.2022.910417
- Amann, T., Hartmann, J., Hellmann, R., Pedrosa, E. T., & Malik, A. (2022). Enhanced weathering potentials—The role of in situ CO2 and grain size distribution. *Frontiers in Climate*, 4, 929268. https://doi.org/10.3389/fclim.2022.929268
- Archer, D. (2005). Fate of fossil fuel CO2 in geologic time. Journal of Geophysical Research, 110(C9), C09S05. https://doi.org/10.1029/ 2004JC002625
- Arribére, M. A., Ribeiro Guevara, S., Sánchez, R. S., Gil, M. I., Román Ross, G., Daurade, L. E., et al. (2003). Heavy metals in the vicinity of a chlor-alkali factory in the upper negro river ecosystem, northern Patagonia, Argentina. Science of the Total Environment, 301(1–3), 187–203. https://doi.org/10.1016/s0048-9697(02)00301-7
- Arts, R., Eiken, O., Chadwick, A., Zweigel, P., van der Meer, L., & Zinszner, B. (2004). Monitoring of CO2 injected at Sleipner using time-lapse seismic data. *Energy*, 29(9), 1383–1392. https://doi.org/10.1016/j.energy.2004.03.072
- Bach, L. T., Gill, S. J., Rickaby, R. E. M., Gore, S., & Renforth, P. (2019). CO2 removal with enhanced weathering and ocean alkalinity enhancement: Potential risks and Co-benefits for marine pelagic ecosystems. *Frontiers in Climate*, 1. https://doi.org/10.3389/fclim.2019.00007 Bach, L. T., Ho, D. T., Boyd, P. W., & Tyka, M. D. (2023). Toward a consensus framework to evaluate air–sea CO2 equilibration for marine CO2
- removal. *Limnology and Oceanography Letters*, 8(5), 685–691. https://doi.org/10.1002/lol2.10330 Bach, L. T., Tamsitt, V., Gower, J., Hurd, C. L., Raven, J. A., & Boyd, P. W. (2021). Testing the climate intervention potential of ocean afforestation using the Great Atlantic Sargassum Belt. *Nature Communications*, 12(1), 2556. https://doi.org/10.1038/s41467-021-22837-2
- Bachu, S. (2015). Review of CO2 storage efficiency in deep saline aquifers. International Journal of Greenhouse Gas Control, 40, 188–202. https://doi.org/10.1016/j.ijggc.2015.01.007
- Baggio, G., Qadir, M., & Smakhtin, V. (2021). Freshwater availability status across countries for human and ecosystem needs. Science of the Total Environment, 792, 148230. https://doi.org/10.1016/j.scitotenv.2021.148230
- Baker, P., Minzlaff, U., Schoenle, A., Schwabe, E., Hohlfeld, M., Jeuck, A., et al. (2018). Potential contribution of surface-dwelling Sargassum algae to deep-sea ecosystems in the southern North Atlantic. Deep-Sea Research Part II: Topical Studies in Oceanography, 148, 21–34. https:// doi.org/10.1016/j.dsr2.2017.10.002
- Bandaranayake, W. M. (1998). Traditional and medicinal uses of mangroves. *Mangroves and Salt Marshes*, 2(3), 133–148. https://doi.org/10. 1023/A:1009988607044
- Bartsch, I., Wiencke, C., Bischof, K., Buchholz, C. M., Buck, B. H., Eggert, A., et al. (2008). The genus Laminaria sensu lato: Recent insights and developments. *European Journal of Phycology*, 43(1), 1–86. https://doi.org/10.1080/09670260701711376
- Baumann, M., Taucher, J., Paul, A. J., Heinemann, M., Vanharanta, M., Bach, L. T., et al. (2021). Effect of intensity and mode of artificial upwelling on particle flux and carbon export. *Frontiers in Marine Science*, 8. https://doi.org/10.3389/fmars.2021.742142
- Berger, M., Kwiatkowski, L., Ho, D. T., & Bopp, L. (2023). Ocean dynamics and biological feedbacks limit the potential of macroalgae carbon dioxide removal. *Environmental Research Letters*, 18(2), 024039. https://doi.org/10.1088/1748-9326/acb06e
- Berner, R. A., Lasaga, A. C., & Garrels, R. M. (1983). The carbonate-silicate geochemical cycle and its effect on atmospheric carbon dioxide over the past 100 million years. American Journal of Science, 283(7), 641–683. https://doi.org/10.2475/ajs.283.7.641
- BGR. (2021). Deutschland rohstoffsituation. Bundesanstalt für Geowissenschaften und Rohstoffe. Retrieved from https://www.bgr.bund.de/DE/ Themen/Min_rohstoffe/Downloads/rohsit-2021.pdf?__blob=publicationFile&v=4
- Bockmon, E. E., & Dickson, A. G. (2015). An inter-laboratory comparison assessing the quality of seawater carbon dioxide measurements. *Marine Chemistry*, 171, 36–43. https://doi.org/10.1016/j.marchem.2015.02.002
- Bolton, J. J., & Lüning, K. (1982). Optimal growth and maximal survival temperatures of Atlantic Laminaria species (Phaeophyta) in culture. *Marine Biology*, 66(1), 89–94. https://doi.org/10.1007/BF00397259
- Borchers, M., Thrän, D., Chi, Y., Dahmen, N., Dittmeyer, R., Dolch, T., et al. (2022). Scoping carbon dioxide removal options for Germany–What is their potential contribution to Net-Zero CO2? *Frontiers in Climate*, *4*. https://doi.org/10.3389/fclim.2022.810343
- Boysen, L. R., Lucht, W., Gerten, D., Heck, V., Lenton, T. M., & Schellnhuber, H. J. (2017). The limits to global-warming mitigation by terrestrial carbon removal. *Earth's Future*, 5(5), 463–474. https://doi.org/10.1002/2016ef000469
- Brunner, C., Hausfather, Z., & Knutti, R. (2024). Durability of carbon dioxide removal is critical for Paris climate goals. *Communications Earth* and Environment, 5(1), 645. https://doi.org/10.1038/s43247-024-01808-7

Acknowledgments

The funding was provided within the framework of the Deutsche Allianz für Meeresforschung (DAM) mission CDRmare and CDRterra. This study was supported by the German Federal Ministry of Education and Research (BMBF) through the following projects: "ASMASYS" (Grant 03F0898), "CDRSynTra" (Grant 01LS2101), "AIMS3" (Grant 03F0964), "BioNET" (Grant 01LS2107A), "STEPSEC" (Grant 01LS2102D), "RETAKE" (Grant 03F0895), "GEOSTORE" (Grant 03F0893), "sea4soCiety" (Grant 03F0896), and "Test-ArtUp" (Grant 03F0897). We also would like to acknowledge the National Key Research and Development Program of China (Grant 2020YFA0608304) and the Emmy Noether scheme by the German Research Foundation 'FOOTPRINTS-From carbOn remOval To achieving the PaRIs agreemeNt's goal: Temperature Stabilisation' (ME 5746/1-1). We thank internal reviewers for valuable input in the manuscript, in particular Wolfgang Koeve. Open Access funding enabled and organized by Projekt DEAL.

- Buck, B. H., & Buchholz, C. M. (2004). The offshore-ring: A new system design for the open ocean aquaculture of macroalgae. *Journal of Applied Phycology*, *16*(5), 355–368. https://doi.org/10.1023/B:JAPH.0000047947.96231.ea
- Buck, B. H., & Buchholz, C. M. (2005). Response of offshore cultivated Laminaria saccharina to hydrodynamic forcing in the North Sea. *Aquaculture*, 250(3), 674–691. https://doi.org/10.1016/j.aquaculture.2005.04.062
- Buck, B. H., Troell, M. F., Krause, G., Angel, D. L., Grote, B., & Chopin, T. (2018). State of the art and challenges for offshore integrated multitrophic aquaculture (IMTA). Frontiers in Marine Science, 5. https://doi.org/10.3389/fmars.2018.00165
- Buck-Wiese, H., Andskog, M. A., Nguyen, N. P., Bligh, M., Asmala, E., Vidal-Melgosa, S., et al. (2023). Fucoid brown algae inject fucoidan carbon into the ocean. Proceedings of the National Academy of Sciences of the United States of America, 120(1), e2210561119. https://doi.org/ 10.1073/pnas.2210561119
- Buylova, A., Fridahl, M., Nasiritousi, N., & Reischl, G. (2021). Cancel (out) emissions? The envisaged role of carbon dioxide removal technologies in long-term national climate strategies. Frontiers in Climate, 3. https://doi.org/10.3389/fclim.2021.675499
- Cameron, C., Hutley, L. B., Friess, D. A., & Brown, B. (2019). High greenhouse gas emissions mitigation benefits from mangrove rehabilitation in Sulawesi, Indonesia. *Ecosystem Services*, 40, 101035. https://doi.org/10.1016/j.ecoser.2019.101035
- Casazza, T. L., & Ross, S. W. (2008). Fishes associated with pelagic Sargassum and open water lacking Sargassum in the Gulf Stream off North Carolina. Retrieved from https://aquadocs.org/handle/1834/25466
- Caserini, S., Barreto, B., Lanfredi, C., Cappello, G., Ross Morrey, D., & Grosso, M. (2019). Affordable CO2 negative emission throughhydrogen from biomass, ocean liming, and CO2storage. *Mitigation and Adaptation Strategies for Global Change*, 24(7), 1231–1248. https://doi.org/10. 1007/s11027-018-9835-7
- Caserini, S., Storni, N., & Grosso, M. (2022). The availability of limestone and other Raw materials for ocean alkalinity enhancement. Global Biogeochemical Cycles, 36(5), e2021GB007246. https://doi.org/10.1029/2021GB007246
- Castro, E., Pinedo, J., Marrugo, J., & León, I. (2022). Retention and vertical distribution of heavy metals in mangrove sediments of the protected area swamp of Mallorquin, Colombian Caribbean. *Regional Studies in Marine Science*, 49, 102072. https://doi.org/10.1016/j.rsma.2021. 102072
- Chatterjee, S., & Huang, K.-W. (2020). Unrealistic energy and materials requirement for direct air capture in deep mitigation pathways. *Nature Communications*, 11(1), 3287. https://doi.org/10.1038/s41467-020-17203-7
- ChemAnalyst. (2023). Decode the future of hydrochloric acid. Retrieved from https://www.chemanalyst.com/industry-report/hydrochloric-acidmarket-707
- Chen, H., Zhou, D., Luo, G., Zhang, S., & Chen, J. (2015). Macroalgae for biofuels production: Progress and perspectives. *Renewable and Sustainable Energy Reviews*, 47, 427–437. https://doi.org/10.1016/j.rser.2015.03.086
- Chen, S., Xu, K., Ji, D., Wang, W., Xu, Y., Chen, C., & Xie, C. (2020). Release of dissolved and particulate organic matter by marine macroalgae and its biogeochemical implications. *Algal Research*, 52, 102096. https://doi.org/10.1016/j.algal.2020.102096
- Chung, I. K., Oak, J. H., Lee, J. A., Shin, J. A., Kim, J. G., & Park, K.-S. (2013). Installing kelp forests/seaweed beds for mitigation and adaptation against global warming: Korean project overview. ICES Journal of Marine Science, 70(5), 1038–1044. https://doi.org/10.1093/icesjms/fss206
- Creutzig, F., Ravindranath, N. H., Berndes, G., Bolwig, S., Bright, R., Cherubini, F., et al. (2015). Bioenergy and climate change mitigation: An assessment. GCB Bioenergy, 7(5), 916–944. https://doi.org/10.1111/gcbb.12205
- Cripps, G., Widdicombe, S., Spicer, J. I., & Findlay, H. S. (2013). Biological impacts of enhanced alkalinity in Carcinus maenas. *Marine Pollution Bulletin*, 71(1–2), 190–198. https://doi.org/10.1016/j.marpolbul.2013.03.015
- Cuba, D., Guardia-Luzon, K., Cevallos, B., Ramos-Larico, S., Neira, E., Pons, A., & Avila-Peltroche, J. (2022). Ecosystem services provided by kelp forests of the humboldt current system: A comprehensive review. *Coasts*, 2(4), 259–277. https://doi.org/10.3390/coasts2040013
- Dean, J., Kiendler-Scharr, A., Mengis, N., Rudich, Y., Schepanski, K., & Zimmermann, R. (2021). Above us only sky. *Communications Earth and Environment*, 2(1), 1–4. https://doi.org/10.1038/s43247-021-00245-0
- Dean, T. A., & Jacobsen, F. R. (1984). Growth of juvenile Macrocystis pyrifera (Laminariales) in relation to environmental factors. *Marine Biology*, 83(3), 301–311. https://doi.org/10.1007/BF00397463
- DEWA. (2022). DEWA annual statistics. Dubai electricity and water authority. Retrieved from https://www.dewa.gov.ae/en/about-us/strategyexcellence/annual-statistics
- Donato, D. C., Kauffman, J. B., Murdiyarso, D., Kurnianto, S., Stidham, M., & Kanninen, M. (2011). Mangroves among the most carbon-rich forests in the tropics. *Nature Geoscience*, 4(5), 293–297. https://doi.org/10.1038/ngeo1123
- Dormann, W. (2023). The surface salinity of the North Sea and Baltic Sea area. In D. Mossakowski & U. Irmler (Eds.), Terrestrial coastal ecosystems in Germany and climate change (pp. 67–73). Springer International Publishing. https://doi.org/10.1007/978-3-031-12539-3_7
- Dowling, A., O'Dwyer, J., & Adley, C. C. (2015). Lime in the limelight. *Journal of Cleaner Production*, 92, 13–22. https://doi.org/10.1016/j. icleoro.2014.12.047
- Duarte, C. M., Losada, I. J., Hendriks, I. E., Mazarrasa, I., & Marbà, N. (2013). The role of coastal plant communities for climate change mitigation and adaptation. *Nature Climate Change*, 3(11), 961–968. https://doi.org/10.1038/nclimate1970

Ebb carbon. (2024). Retrieved February 20, 2024, from https://www.ebbcarbon.com

- Eger, A. M., Marzinelli, E. M., Beas-Luna, R., Blain, C. O., Blamey, L. K., Byrnes, J. E. K., et al. (2023). The value of ecosystem services in global marine kelp forests. *Nature Communications*, 14(1), 1894. https://doi.org/10.1038/s41467-023-37385-0
- Eisaman, M. D., Geilert, S., Renforth, P., Bastianini, L., Campbell, J., Dale, A. W., et al. (2023). Assessing the technical aspects of oceanalkalinity-enhancement approaches. *State of the Planet*, 2-oae2023, 1–29. https://doi.org/10.5194/sp-2-oae2023-3-2023
- EuLA. (2014). A competitive and efficient lime industry. The European Lime Association. Retrieved from https://www.eula.eu/wp-content/ uploads/2019/02/A-Competitive-and-Efficient-Lime-Industry-Presentation.pdf
- European Commission. (2014). Technology readiness levels (TRL). In *Horizon 2020—WORK PROGRAMME 2014-2015 general annexes,* extract from Part 19—commission decision C. European Commission Brussels. Retrieved from https://ec.europa.eu/research/participants/data/ ref/h2020/wp/2014_2015/annexes/h2020-wp1415-annex-g-trl_en.pdf
- Falkowski, P., Scholes, R. J., Boyle, E., Canadell, J., Canfield, D., Elser, J., et al. (2000). The global carbon cycle: A test of our knowledge of earth as a system. *Science*, 290(5490), 291–296. https://doi.org/10.1126/science.290.5490.291
- Federal Climate Change Act. (2019). § Federal Law Gazette, I. Retrieved from https://www.gesetze-im-internet.de/ksg/
- Ferderer, A., Chase, Z., Kennedy, F., Schulz, K. G., & Bach, L. T. (2022). Assessing the influence of ocean alkalinity enhancement on a coastal phytoplankton community. *Biogeosciences*, 19(23), 5375–5399. https://doi.org/10.5194/bg-19-5375-2022
- Fernand, F., Israel, A., Skjermo, J., Wichard, T., Timmermans, K. R., & Golberg, A. (2017). Offshore macroalgae biomass for bioenergy production: Environmental aspects, technological achievements and challenges. *Renewable and Sustainable Energy Reviews*, 75, 35–45. https:// doi.org/10.1016/j.rser.2016.10.046

23284277, 2025, 4, Downloaded from

- Filbee-Dexter, K. (2020). Ocean forests hold unique solutions to our current environmental crisis. One Earth, 2(5), 398-401. https://doi.org/10. 1016/j.oneear.2020.05.004
- Flipkens, G., Fuhr, M., Fiers, G., Meysman, F. J. R., Town, R. M., & Blust, R. (2023). Enhanced olivine dissolution in seawater through continuous grain collisions. *Geochimica et Cosmochimica Acta*, 359, 84–99. https://doi.org/10.1016/j.gca.2023.09.002
- Fortes, M. D., & Lüning, K. (1980). Growth rates of North Sea macroalgae in relation to temperature, irradiance and photoperiod. *Helgolaender Meeresuntersuchungen*, 34(1), 15–29. https://doi.org/10.1007/BF01983538
- Foteinis, S., Andresen, J., Campo, F., Caserini, S., & Renforth, P. (2022). Life cycle assessment of ocean liming for carbon dioxide removal from the atmosphere. *Journal of Cleaner Production*, 370, 133309. https://doi.org/10.1016/J.JCLEPRO.2022.133309
- Fourqurean, J. W., Duarte, C. M., Kennedy, H., Marbà, N., Holmer, M., Mateo, M. A., et al. (2012). Seagrass ecosystems as a globally significant carbon stock. *Nature Geoscience*, 5(7), 505–509. https://doi.org/10.1038/ngeo1477
- Fredriksen, S., Filbee-Dexter, K., Norderhaug, K. M., Steen, H., Bodvin, T., Coleman, M. A., et al. (2020). Green gravel: A novel restoration tool to combat kelp forest decline. *Scientific Reports*, 10(1), 3983. https://doi.org/10.1038/s41598-020-60553-x
- Fridahl, M., Hansson, A., & Haikola, S. (2020). Towards indicators for a negative emissions climate stabilisation index: Problems and prospects. *Climate*, 8(6), 75. https://doi.org/10.3390/cli8060075
- Fuhr, M., Wallmann, K., Dale, A. W., Kalapurakkal, H. T., Schmidt, M., Sommer, S., et al. (2024). Alkaline mineral addition to anoxic to hypoxic Baltic Sea sediments as a potentially efficient CO2-removal technique. *Frontiers in Climate*, 6, 1338556. https://doi.org/10.3389/fclim.2024. 1338556
- Fujimori, S., Wu, W., Doelman, J., Frank, S., Hristov, J., Kyle, P., et al. (2022). Land-based climate change mitigation measures can affect agricultural markets and food security. *Nature Food*, 3(2), 110–121. https://doi.org/10.1038/s43016-022-00464-4
- Fuss, S., Canadell, J. G., Peters, G. P., Tavoni, M., Andrew, R. M., Ciais, P., et al. (2014). Betting on negative emissions. *Nature Climate Change*, 4(10), 850–853. https://doi.org/10.1038/nclimate2392
- Fuss, S., Lamb, W. F., Callaghan, M. W., Hilaire, J., Creutzig, F., Amann, T., et al. (2018). Negative emissions Part 2: Costs, potentials and side effects. *Environmental Research Letters*, 13(6), 063002. https://doi.org/10.1088/1748-9326/aabf9f
- Gao, G., Gao, L., Jiang, M., Jian, A., & He, L. (2021). The potential of seaweed cultivation to achieve carbon neutrality and mitigate deoxygenation and eutrophication. *Environmental Research Letters*, 17(1), 014018. https://doi.org/10.1088/1748-9326/ac3fd9
- Gately, J. A., Kim, S. M., Jin, B., Brzezinski, M. A., & Iglesias-Rodriguez, M. D. (2023). Coccolithophores and diatoms resilient to ocean alkalinity enhancement: A glimpse of hope? *Science Advances*, 9(24), eadg6066. https://doi.org/10.1126/sciadv.adg6066
- Gattuso, J.-P., Magnan, A. K., Bopp, L., Cheung, W. W. L., Duarte, C. M., Hinkel, J., et al. (2018). Ocean solutions to address climate change and its effects on marine ecosystems. *Frontiers in Marine Science*, *5*, 337. https://doi.org/10.3389/fmars.2018.00337
- Gattuso, J.-P., Williamson, P., Duarte, C. M., & Magnan, A. K. (2021). The potential for ocean-based climate action: Negative emissions technologies and beyond. *Frontiers in Climate*, 2(January), 1–8. https://doi.org/10.3389/fclim.2020.575716
- Gerbens-Leenes, P. W., Hoekstra, A. Y., & Bosman, R. (2018). The blue and grey water footprint of construction materials: Steel, cement and glass. *Water Resources and Industry*, 19, 1–12. https://doi.org/10.1016/j.wri.2017.11.002
- Ginzky, H., Herrmann, F., Kartschall, K., Leujak, W., Lipsius, K., Maeder, C., et al. (2011). Geo-engineering: Effective climate protection or megalomania? Retrieved from https://www.osti.gov/etdeweb/biblio/21462852
- Giri, C., Ochieng, E., Tieszen, L. L., Zhu, Z., Singh, A., Loveland, T., et al. (2011). Status and distribution of mangrove forests of the world using earth observation satellite data. *Global Ecology and Biogeography*, 20(1), 154–159. https://doi.org/10.1111/j.1466-8238.2010.00584.x
- Goertz-Allmann, B. P., Kühn, D., Oye, V., Bohloli, B., & Aker, E. (2014). Combining microseismic and geomechanical observations to interpret storage integrity at the in Salah CCS site. *Geophysical Journal International*, 198(1), 447–461. https://doi.org/10.1093/gji/ggu010
- Gouvêa, L. P., Assis, J., Gurgel, C. F. D., Serrão, E. A., Silveira, T. C. L., Santos, R., et al. (2020). Golden carbon of Sargassum forests revealed as an opportunity for climate change mitigation. Science of the Total Environment, 729, 138745. https://doi.org/10.1016/j.scitotenv.2020.138745
- Grabemann, I., & Weisse, R. (2008). Climate change impact on extreme wave conditions in the North Sea: An ensemble study. *Ocean Dynamics*, 58(3), 199–212. https://doi.org/10.1007/s10236-008-0141-x
- Hangx, S. J. T., & Spiers, C. J. (2009). Coastal spreading of olivine to control atmospheric CO2 concentrations: A critical analysis of viability. International Journal of Greenhouse Gas Control, 3(6), 757–767. https://doi.org/10.1016/j.ijggc.2009.07.001
- Hartmann, J., Suitner, N., Lim, C., Schneider, J., Marín-Samper, L., Arístegui, J., et al. (2023). Stability of alkalinity in ocean alkalinity enhancement (OAE) approaches – Consequences for durability of CO2 storage. *Biogeosciences*, 20(4), 781–802. https://doi.org/10.5194/bg-20-781-2023
- Hauck, J., Köhler, P., Wolf-Gladrow, D., & Völker, C. (2016). Iron fertilisation and century-scale effects of open ocean dissolution of olivine in a simulated CO2 removal experiment. *Environmental Research Letters*, 11(2), 024007. https://doi.org/10.1088/1748-9326/11/2/024007
- Heck, V., Gerten, D., Lucht, W., & Popp, A. (2018). Biomass-based negative emissions difficult to reconcile with planetary boundaries. Nature Climate Change, 8(2), 151–155. https://doi.org/10.1038/s41558-017-0064-y
- Heyward, C. (2019). 21 normative issues of geoengineering technologies. In T. M. Letcher (Ed.), *Managing global warming* (pp. 639–657). Academic Press. https://doi.org/10.1016/B978-0-12-814104-5.00021-1
- Heß, D., Klumpp, M., & Dittmeyer, R. (2020). Nutzung von CO2 aus Luft als Rohstoff für synthetische Kraftstoffe und Chemikalien. Karlsruhe Institute of Technology (KIT). Retrieved from https://vm.baden-wuerttemberg.de/fileadmin/redaktion/m-mvi/intern/Dateien/PDF/29-01-2021-DAC-Studie.pdf
- Hinrichs, C., Köhler, P., Völker, C., & Hauck, J. (2023). Alkalinity biases in CMIP6 Earth system models and implications for simulated CO 2 drawdown via artificial alkalinity enhancement. *Biogeosciences*, 20(18), 3717–3735. https://doi.org/10.5194/bg-20-3717-2023
- Ho, D. T., Bopp, L., Palter, J. B., Long, M. C., Boyd, P. W., Neukermans, G., & Bach, L. T. (2023). Monitoring, reporting, and verification for ocean alkalinity enhancement. *Guide to Best Practices in Ocean Alkalinity Enhancement Research*, 2-oae2023, 1–12. 2-oae2023, 12. https:// doi.org/10.5194/sp-2-oae2023-12-2023
- Hoppema, J. M. J. (1990). The distribution and seasonal variation of alkalinity in the southern Bight of the North Sea and in the Western Wadden sea. *Netherlands Journal of Sea Research*, 26(1), 11–23. https://doi.org/10.1016/0077-7579(90)90053-J
- House, K. Z., House, C. H., Schrag, D. P., & Aziz, M. J. (2007). Electrochemical acceleration of chemical weathering as an energetically feasible approach to mitigating anthropogenic climate change. *Environmental Science and Technology*, 41(24), 8464–8470. https://doi.org/10.1021/ es0701816
- Hsiung, A. R., Ong, O. X. J., Teo, X. S., Friess, D. A., Todd, P. A., Swearer, S. E., & Morris, R. L. (2024). Determinants of mangrove seedling survival incorporated within hybrid living shorelines. *Ecological Engineering*, 202, 107235. https://doi.org/10.1016/j.ecoleng.2024.107235
- Hutchins, D. A., Fu, F.-X., Yang, S.-C., John, S. G., Romaniello, S. J., Andrews, M. G., & Walworth, N. G. (2023). Responses of globally important phytoplankton species to olivine dissolution products and implications for carbon dioxide removal via ocean alkalinity enhancement. *Biogeosciences*, 20(22), 4669–4682. https://doi.org/10.5194/bg-20-4669-2023

- Hylén, A., Kreuzburg, M., De Wolf, S., Burdorf, L., Fiers, G., Goossens, C., et al. (2023). Ocean alkalinity enhancement through enhanced silicate weathering in coastal areas: A long-term mesocosm study. https://doi.org/10.5194/egusphere-egu23-14128
- IEA. (2022). Direct air capture: A key technology for net zero. OECD. https://doi.org/10.1787/bbd20707-en
- IEA. (2023). United Arab Emirates renewables. February 21, 2024, Retrieved from https://www.iea.org/countries/united-arab-emirates

Ilyina, T., Wolf-Gladrow, D., Munhoven, G., & Heinze, C. (2013). Assessing the potential of calcium-based artificial ocean alkalinization to mitigate rising atmospheric CO2 and ocean acidification. *Geophysical Research Letters*, 40(22), 5909–5914. https://doi.org/10.1002/ 2013GL057981

IPCC. (2022). Global warming of 1.5°C. Global warming of 1.5°C. Cambridge University Press. https://doi.org/10.1017/9781009157940

- IPCC. (2023). In Climate change 2022 mitigation of climate change: Working group III contribution to the sixth assessment report of the intergovernmental Panel on climate change (1st ed.). Cambridge University Press. https://doi.org/10.1017/9781009157926
- Jamesh, M.-I., & Harb, M. (2021). Tuning the electronic structure of the earth-abundant electrocatalysts for oxygen evolution reaction (OER) to achieve efficient alkaline water splitting – a review. Journal of Materials Chemistry A: Materials for Energy and Sustainability, 56, 299–342. https://doi.org/10.1016/j.jechem.2020.08.001
- Jia, Y., Quack, B., Kinley, R. D., Pisso, I., & Tegtmeier, S. (2022). Potential environmental impact of bromoform from Asparagopsis farming in Australia. Atmospheric Chemistry and Physics, 22(11), 7631–7646. https://doi.org/10.5194/acp-22-7631-2022
- Johnson, D. L., & Richardson, P. L. (1977). On the wind-induced sinking of Sargassum. Journal of Experimental Marine Biology and Ecology, 28(3), 255–267. https://doi.org/10.1016/0022-0981(77)90095-8
- Jones, D. C., Ito, T., Takano, Y., & Hsu, W. (2014). Spatial and seasonal variability of the air-sea equilibration timescale of carbon dioxide. *Global Biogeochemical Cycles*, 28(11), 1163–1178. https://doi.org/10.1002/2014GB004813
- Jones, E., Qadir, M., Van Vliet, M. T. H., Smakhtin, V., & Kang, S. (2019). The state of desalination and brine production: A global outlook. Science of the Total Environment, 657, 1343–1356. https://doi.org/10.1016/j.scitotenv.2018.12.076
- Jürchott, M., Oschlies, A., & Koeve, W. (2023). Artificial upwelling—A refined narrative. Geophysical Research Letters, 50(4). https://doi.org/ 10.1029/2022g1101870
- Kayalvizhi, K., & Kathiresan, K. (2019). Microbes from wastewater treated mangrove soil and their heavy metal accumulation and Zn solubilization. *Biocatalysis and Agricultural Biotechnology*, 22, 101379. https://doi.org/10.1016/j.bcab.2019.101379
- Kelemen, P., Benson, S. M., Pilorgé, H., Psarras, P., & Wilcox, J. (2019). An overview of the status and challenges of CO2 storage in minerals and geological formations. *Frontiers in Climate*, 1. https://doi.org/10.3389/fclim.2019.00009
- Keller, D. P., Feng, E. Y., & Oschlies, A. (2014). Potential climate engineering effectiveness and side effects during a high carbon dioxideemission scenario. *Nature Communications*, 5, 1–11. https://doi.org/10.1038/ncomms4304
- Kemper, J., Mense, J., Graf, K., Riebesell, U., & Kröger, J. (2023). Towards reliable performance predictions for stommel's perpetual salt fountain. In *Presented at the 10th conference on computational methods in marine engineering*. CIMNE. https://doi.org/10.23967/marine. 2023.018
- Keng, F. S.-L., Phang, S.-M., Rahman, N. A., Leedham, E. C., Hughes, C., Robinson, A. D., et al. (2013). Volatile halocarbon emissions by three tropical brown seaweeds under different irradiances. *Journal of Applied Phycology*, 25(5), 1377–1386. https://doi.org/10.1007/s10811-013-9990-x
- Kim, J. K., Yarish, C., Hwang, E. K., Park, M., Kim, Y., Kim, J. K., et al. (2017). Seaweed aquaculture: Cultivation technologies, challenges and its ecosystem services. *Algae*, 32(1), 1–13. https://doi.org/10.4490/algae.2017.32.3.3
- Klepper, G., & Rickels, W. (2012). The real economics of climate engineering. *Economics Research International*, 2012, 1–20. https://doi.org/10. 1155/2012/316564
- Knopf, S., & May, F. (2017). Comparing methods for the estimation of CO2 storage capacity in saline aquifers in Germany: Regional aquifer based vs. Structural trap based assessments. *Energy Proceedia*, 114, 4710–4721. https://doi.org/10.1016/j.egypro.2017.03.1605
- Köhler, P. (2020). Anthropogenic CO2 of high emission scenario compensated after 3500 years of ocean alkalinization with an annually constant dissolution of 5 Pg of olivine. Frontiers in Climate, 2, 575744. https://doi.org/10.3389/fclim.2020.575744
- Köhler, P., Abrams, J. F., Völker, C., Hauck, J., & Wolf-Gladrow, D. A. (2013). Geoengineering impact of open ocean dissolution of olivine on atmospheric CO2, surface ocean pH and marine biology. *Environmental Research Letters*, 8(1), 014009. https://doi.org/10.1088/1748-9326/8/ 1/014009
- Krause-Jensen, D., & Duarte, C. M. (2016). Substantial role of macroalgae in marine carbon sequestration. Nature Publishing Group. https://doi. org/10.1038/NGE02790
- Kwiatkowski, L., Ricke, K. L., & Caldeira, K. (2015). Atmospheric consequences of disruption of the ocean thermocline. *Environmental Research Letters*, 10(3), 034016. https://doi.org/10.1088/1748-9326/10/3/034016
- Lakshmanan, S., & Murugesan, T. (2014). The chlor-alkali process: Work in Progress. Clean Technologies and Environmental Policy, 16(2), 225–234. https://doi.org/10.1007/s10098-013-0630-6
- Langer, G., Biogeosciences, M. G., Oceanography, B., Wegener, A., Baumann, K.-H., Kläs, J., & Young, J. R. (2006). Species-specific responses of calcifying algae to changing seawater carbonate chemistry. *Geochemistry, Geophysics, Geosystems*, 7(9). https://doi.org/10.1029/ 2005GC001227
- Lattemann, S., & Höpner, T. (2008). Environmental impact and impact assessment of seawater desalination. *Desalination*, 220(1), 1–15. https://doi.org/10.1016/j.desal.2007.03.009
- Lenzi, D., Lamb, W. F., Hilaire, J., Kowarsch, M., & Minx, J. C. (2018). Don't deploy negative emissions technologies without ethical analysis. *Nature*, 561(7723), 303–305. https://doi.org/10.1038/d41586-018-06695-5
- Levin, L. A., Alfaro-Lucas, J. M., Colaço, A., Cordes, E. E., Craik, N., Danovaro, R., et al. (2023). Deep-sea impacts of climate interventions. *Science*, 379(6636), 978–981. https://doi.org/10.1126/science.ade7521
- Locke, A., Doe, K., Fairchild, W., Jackman, P., Reese, E., & Carman, M. (2009). Preliminary evaluation of effects of invasive tunicate management with acetic acid and calcium hydroxide on non-target marine organisms in Prince Edward Island, Canada. Aquatic Invasions, 4(1), 221–236. https://doi.org/10.3391/ai.2009.4.1.23
- Luderer, G., Günther, C., Sörgel, D., Kost, C., Benke, F., Auer, C., et al. (2021). Deutschland auf dem Weg zur Klimaneutralität 2045 Szenarien und Pfade im Modellvergleich.
- Lumley, D. E. (2001). Time-lapse seismic reservoir monitoring. Geophysics, 66(1), 50-53. https://doi.org/10.1190/1.1444921
- Macovei, V. A., Petersen, W., Brix, H., & Voynova, Y. G. (2021). Reduced ocean carbon sink in the south and central North Sea (2014–2018) revealed from FerryBox observations. *Geophysical Research Letters*, 48(11). https://doi.org/10.1029/2021g1092645

Ilman, M., Dargusch, P., & Dart, P., & Onrizal. (2016). A historical analysis of the drivers of loss and degradation of Indonesia's mangroves. Land Use Policy, 54, 448–459. https://doi.org/10.1016/j.landusepol.2016.03.010

- Macreadie, P. I., Costa, M. D. P., Atwood, T. B., Friess, D. A., Kelleway, J. J., Kennedy, H., et al. (2021). Blue carbon as a natural climate solution. *Nature Reviews Earth and Environment*, 2(12), 826–839. https://doi.org/10.1038/s43017-021-00224-1
- Mammoth. (2022). Retrieved February 23, 2024, from https://climeworks.com/news/climeworks-announces-groundbreaking-on-mammoth Marappan, S., Stokke, R., Malinovsky, M. P., & Taylor, A. (2022). Assessment of impacts of the offshore oil and gas industry on the marine environment. OSPAR Commission. Retrieved from https://oap.ospar.org/en/ospar-assessments/quality-status-reports/qsr-2023/other-
- assessments/impacts-offshore-oil-and-gas-industry/ Martin, L. M., Taylor, M., Huston, G., Goodwin, D. S., Schell, J. M., & Siuda, A. N. S. (2021). Pelagic Sargassum morphotypes support different
- rating motile epifauna communities. *Marine Biology*, 168(7), 115. https://doi.org/10.1007/s0227-021-03910-2
- McLaren, D. (2012). A comparative global assessment of potential negative emissions technologies. Process Safety and Environmental Protection, 90(6), 489–500. https://doi.org/10.1016/J.PSEP.2012.10.005
- Mengis, N., Kalhori, A., Simon, S., Harpprecht, C., Baetcke, L., Prats-Salvado, E., et al. (2022). Net-zero CO 2 Germany—A retrospect from the year 2050. *Earth's Future*, *10*(2), 1–16. https://doi.org/10.1029/2021EF002324
- Mengis, N., Paul, A., & Fernández-Méndez, M. (2023). Counting (on) blue carbon—Challenges and ways forward for carbon accounting of ecosystem-based carbon removal in marine environments. *PLOS Climate*, 2(8), e0000148. https://doi.org/10.1371/journal.pclm.0000148
- Merfort, A., Stevanović, M., & Strefler, J. (2023). Energiewende auf Netto-Null: Passen Angebot und Nachfrage nach CO2 -Entnahme aus der Atmosphäre zusammen? Kopernikus-Projekt Ariadne.
- Merk, C., Grunau, J., Riekhof, M.-C., & Rickels, W. (2022). The need for local governance of global commons: The example of blue carbon ecosystems. *Ecological Economics*, 201, 107581. https://doi.org/10.1016/j.ecolecon.2022.107581
- Meysman, F. J. R., & Montserrat, F. (2017). Negative CO 2 emissions via enhanced silicate weathering in coastal environments. *Biology Letters*, 13(4), 20160905. https://doi.org/10.1098/rsbl.2016.0905
- Mithoo-Singh, P. K., Keng, F. S.-L., Phang, S.-M., Leedham Elvidge, E. C., Sturges, W. T., Malin, G., & Abd Rahman, N. (2017). Halocarbon emissions by selected tropical seaweeds: Species-specific and compound-specific responses under changing pH. *PeerJ*, 5, e2918. https://doi. org/10.7717/peerj.2918
- Moosdorf, N., Renforth, P., & Hartmann, J. (2014). Carbon dioxide efficiency of terrestrial enhanced weathering. *Environmental Science and Technology*, 48(9), 4809–4816. https://doi.org/10.1021/es4052022
- Moras, C. A., Bach, L. T., Cyronak, T., Joannes-Boyau, R., & Schulz, K. G. (2022). Ocean alkalinity enhancement Avoiding runaway CaCO3 precipitation during quick and hydrated lime dissolution. *Biogeosciences*, 19(15), 3537–3557. https://doi.org/10.5194/bg-19-3537-2022
- Murphy, J. D., Drosg, B., Allen, E., Jerney, J., Xia, A., & Herrmann, C. (2015). A perspective on algal biogas. IEA Bioeconomy. Retrieved from https://www.ieabioenergy.com/blog/publications/a-perspective-on-algal-biogas/
- Nanz, P., & Fritsche, M. (2012). Handbuch Bürgerbeteiligung: Verfahren und Akteure, Chancen und Grenzen. Bundeszentrale für Politische Bildung. Retrieved from https://play.google.com/store/books/details?id=gCkSMwEACAAJ
- NAS. (2021). A research strategy for ocean-based carbon dioxide removal and sequestration. National Academies Press (US). https://doi.org/10. 17226/26278
- Norström, A. V., Cvitanovic, C., Löf, M. F., West, S., Wyborn, C., Balvanera, P., et al. (2020). Principles for knowledge co-production in sustainability research. *Nature Sustainability*, 3(3), 182–190. https://doi.org/10.1038/s41893-019-0448-2
- Orca. (2021). Retrieved February 23, 2024, from https://climeworks.com/press-release/climeworks-launches-orca
- Ortiz, J., Arístegui, J., Hernández-Hernández, N., Fernández-Méndez, M., & Riebesell, U. (2022). Oligotrophic phytoplankton community effectively adjusts to artificial upwelling regardless of intensity, but differently among upwelling modes. *Frontiers in Marine Science*, 9. https://doi.org/10.3389/fmars.2022.880550
- Oschlies, A., Held, H., Keller, D., Keller, K., Mengis, N., Quaas, M., et al. (2017). Indicators and metrics for the assessment of climate engineering. *Earth's Future*, 5(1), 49–58. https://doi.org/10.1002/2016EF000449
- Oschlies, A., Koeve, W., Rickels, W., & Rehdanz, K. (2010). Side effects and accounting aspects of hypothetical large-scale Southern Ocean iron fertilization. *Biogeosciences*, 7(12), 4014–4035. https://doi.org/10.5194/bg-7-4017-2010
- Oschlies, A., Pahlow, M., Yool, A., & Matear, R. J. (2010). Climate engineering by artificial ocean upwelling: Channelling the sorcerer's apprentice. *Geophysical Research Letters*, 37(4). https://doi.org/10.1029/2009GL041961
- Oschlies, A., Stevenson, A., Bach, L. T., Fennel, K., Rickaby, R. E. M., Satterfield, T., et al. (2023). Guide to best practices in ocean alkalinity enhancement research (0 ed.). Copernicus GmbH. https://doi.org/10.5194/sp-2-oae2023
- Pedersen, F. M., & Hansen, P. J. (2003). Effects of high pH on a natural marine planktonic community. *Marine Ecology Progress Series*, 260, 19–31. https://doi.org/10.3354/meps260019
- Pedersen, M. F., Filbee-Dexter, K., Frisk, N. L., Sárossy, Z., & Wernberg, T. (2021). Carbon sequestration potential increased by incomplete anaerobic decomposition of kelp detritus. *Marine Ecology Progress Series*, 660, 53–67. https://doi.org/10.3354/meps13613
- Planke, S., Bellwald, B., Millett, J., Planke, E. E. E., Zastrozhnov, D., Carlevaris, P., et al. (2021). Permanent carbon sequestration potential in offshore basalt sequences on the NW European continental margins. In *Presented at the 82nd EAGE annual conference and exhibition*. European Association of Geoscientists and Engineers. https://doi.org/10.3997/2214-4609.202011841
- Planke, S., Berndt, C., & Alvarez Zarikian, C. A. (2022). Expedition 396 preliminary report: Mid-Norwegian margin magmatism and paleoclimate implications. *International Ocean Discovery Program*, 396. https://doi.org/10.14379/iodp.pr.396.2022
- Polidoro, B. A., Carpenter, K. E., Collins, L., Duke, N. C., Ellison, A. M., Ellison, J. C., et al. (2010). The loss of species: Mangrove extinction risk and geographic areas of global concern. PLoS One, 5(4), e10095. https://doi.org/10.1371/journal.pone.0010095
- Powers, L. C., Hertkorn, N., McDonald, N., Schmitt-Kopplin, P., Del Vecchio, R., Blough, N. V., & Gonsior, M. (2019). Sargassum sp. Act as a large regional source of marine dissolved organic carbon and polyphenols. *Global Biogeochemical Cycles*, 33(11), 1423–1439. https://doi.org/ 10.1029/2019gb006225
- Preston, C. J. (2013). Ethics and geoengineering: Reviewing the moral issues raised by solar radiation management and carbon dioxide removal. *WIREs Climate Change*, 4(1), 23–37. https://doi.org/10.1002/wcc.198
- Ramasamy, M., Amann, T., & Moosdorf, N. (2024). Regional potential of coastal ocean alkalinization with olivine within 100 years. Environmental Research Letters, 19(6), 064030. https://doi.org/10.1088/1748-9326/ad4664
- Rau, G. H., Carroll, S. A., Bourcier, W. L., Singleton, M. J., Smith, M. M., & Aines, R. D. (2013). Direct electrolytic dissolution of silicate minerals for air CO2 mitigation and carbon-negative H2 production. *Proceedings of the National Academy of Sciences*, 110(25), 10095–10100. https://doi.org/10.1073/pnas.1222358110
- Rau, G. H., Willauer, H. D., & Ren, Z. J. (2018). The global potential for converting renewable electricity to negative-CO2-emissions hydrogen. *Nature Climate Change*, 8(7), 621–625. https://doi.org/10.1038/s41558-018-0203-0

- Raw, J. L., Van Niekerk, L., Chauke, O., Mbatha, H., Riddin, T., & Adams, J. B. (2023). Blue carbon sinks in South Africa and the need for restoration to enhance carbon sequestration. *Science of the Total Environment*, 859(Pt 1), 160142. https://doi.org/10.1016/j.scitotenv.2022. 160142
- Realmonte, G., Drouet, L., Gambhir, A., Glynn, J., Hawkes, A., Köberle, A. C., & Tavoni, M. (2019). An inter-model assessment of the role of direct air capture in deep mitigation pathways. *Nature Communications*, 10(1), 3277. https://doi.org/10.1038/s41467-019-10842-5
- Realmonte, G., Drouet, L., Gambhir, A., Glynn, J., Hawkes, A., Köberle, A. C., & Tavoni, M. (2020). Reply to "High energy and materials requirement for direct air capture calls for further analysis and R&D.". *Nature Communications*, 11(1), 3286. https://doi.org/10.1038/s41467-020-17204-6
- Renforth, P., & Henderson, G. (2017). Assessing ocean alkalinity for carbon sequestration. *Reviews of Geophysics*, 55(3), 636–674. https://doi.org/10.1002/2016RG000533
- Rickels, W. (2023). Kapitel 6.6: Wirtschaftliche Aspekte bei der atmosphärischen CO2-Entnahme und dem Strahlungsmanagement. In *Warning signal climate: Does technology help against global warming?* (pp. 291–297). Retrieved from https://www.ifw-kiel.de/de/publikationen/kapitel-66-wirtschaftliche-aspekte-bei-der-atmosphaerischen-co2-entnahme-und-dem-strahlungsmanagement-31823/
- Roberts, D. A., Johnston, E. L., & Knott, N. A. (2010). Impacts of desalination plant discharges on the marine environment: A critical review of published studies. Water Research, 44(18), 5117–5128. https://doi.org/10.1016/j.watres.2010.04.036
- Roesijadi, G., Jones, S. B., Snowden-Swan, L. J., & Zhu, Y. (2010). Macroalgae as a biomass feedstock: A preliminary analysis. https://doi.org/10. 2172/1006310
- Rosenqvist, M. P., Meakins, M. W. J., Planke, S., Millett, J. M., Kjøll, H. J., Voigt, M. J., & Jamtveit, B. (2023). Reservoir properties and reactivity of the Faroe Islands basalt group: Investigating the potential for CO2 storage in the North Atlantic Igneous Province. *International Journal of Greenhouse Gas Control*, 123, 103838. https://doi.org/10.1016/j.ijggc.2023.103838 Sarga agriscience. (2021a). Retrieved November 2, 2023, from https://sarga.ag/
- Sarga agriscience. (2021a). Retrieved November 2, 2023, from https://sarga.ag/
- Sasmito, S. D., Basyuni, M., Kridalaksana, A., Saragi-Sasmito, M. F., Lovelock, C. E., & Murdiyarso, D. (2023). Challenges and opportunities for achieving Sustainable Development Goals through restoration of Indonesia's mangroves. *Nature Ecology and Evolution*, 7(1), 62–70. https:// doi.org/10.1038/s41559-022-01926-5
- Satterfield, T., Nawaz, S., & Boettcher, M. (2023). Social considerations and best practices to apply to engaging publics on ocean alkalinity enhancement. *Guide to Best Practices in Ocean Alkalinity Enhancement Research*, *11*, 2–oae2023. https://doi.org/10.5194/sp-2-oae2023-11-2023
- Segreto, M., Principe, L., Desormeaux, A., Torre, M., Tomassetti, L., Tratzi, P., et al. (2020). Trends in social acceptance of renewable energy across Europe-A literature review. *International Journal of Environmental Research and Public Health*, 17(24), 9161. https://doi.org/10.3390/ ijerph17249161
- Sengupta, M. (2021). Environmental impacts of mining: Monitoring, restoration, and control. Taylor and Francis Group. https://doi.org/10.1201/ 9781003164012
- Siegel, D. A., DeVries, T., Doney, S. C., & Bell, T. (2021). Assessing the sequestration time scales of some ocean-based carbon dioxide reduction strategies. *Environmental Research Letters*, 16(10), 104003. https://doi.org/10.1088/1748-9326/ac0be0
- Smith, S. M., Geden, O., Nemet, G. F., Gidden, M. J., Lamb, W. F., Powis, C., et al. (2023). The state of carbon dioxide removal 1st edition. The state of carbon dioxide removal. https://doi.org/10.17605/OSF.IO/W3B4Z
- Snæbjörnsdóttir, S. Ó., & Gislason, S. R. (2016). CO2 storage potential of basaltic rocks offshore Iceland. Energy Procedia, 86, 371–380. https:// doi.org/10.1016/j.egypro.2016.01.038
- Sonnichsen, C., Atamanchuk, D., Hendricks, A., Morgan, S., Smith, J., Grundke, I., et al. (2023). An automated microfluidic analyzer for in situ monitoring of total alkalinity. ACS Sensors, 8(1), 344–352. https://doi.org/10.1021/acssensors.2c02343
- Sterr, H. (2008). Assessment of vulnerability and adaptation to sea-level rise for the coastal zone of Germany. Journal of Coastal Research, 242, 380–393. https://doi.org/10.2112/07A-0011.1
- Stevenson, A., Ó Corcora, T. C., Hukriede, W., Schubert, P. R., & Reusch, T. B. H. (2023). Substantial seagrass blue carbon pools in the southwestern Baltic Sea include relics of terrestrial peatlands. *Frontiers in Marine Science*, 10, 1266663. https://doi.org/10.3389/fmars.2023. 1266663
- Stommel, H., Arons, A. B., & Blanchard, D. (1956). An oceanographical curiosity: The perpetual salt fountain. Deep-Sea Research, 3(2), 152– 153. https://doi.org/10.1016/0146-6313(56)90095-8

Sündermann, J., & Pohlmann, T. (2011). A brief analysis of North Sea physics. *Oceanologia*, 53(3), 663–689. https://doi.org/10.5697/oc.53-3.663 Suutari, M., Leskinen, E., Fagerstedt, K., Kuparinen, J., Kuuppo, P., & Blomster, J. (2015). Macroalgae in biofuel production. *Phycological*

- Suutari, M., Leskinen, E., Fagerstedt, K., Kuparinen, J., Kuuppo, P., & Blomster, J. (2015). Macroalgae in biofuel production. *Phycologic Research*, 63(1), 1–18. https://doi.org/10.1111/pre.12078
- Tames-Espinosa, M., Martínez, I., Romero-Kutzner, V., Coca, J., Algueró-Muñiz, M., Horn, H. G., et al. (2020). Metabolic responses of subtropical microplankton after a simulated deep-water upwelling event suggest a possible dominance of mixotrophy under increasing CO2 levels. *Frontiers in Marine Science*, 7, 307. https://doi.org/10.3389/fmars.2020.00307
- Tank, L., Voget-Kleschin, L., Garschagen, M., Boettcher, M., Mengis, N., Holland-Cunz, A., et al. (2025). Distinguish between feasibility and desirability when assessing climate response options. *Npj Climate Action*, 4(1). https://doi.org/10.1038/s44168-025-00237-2
- Tedsen, E., & Homann, G. (2013). Implementing the precautionary principle for climate engineering. *Carbon and Climate Law Review*, 7(2), 90–100. https://doi.org/10.21552/cclr/2013/2/250
- Tennet. (2023). Nordsee-,,Windernte" 2022 steigt um vier Prozent gegenüber dem Vorjahr auf gut 21 Terawattstunden. November 12, 2023, Retrieved from https://www.tennet.eu/de/news/nordsee-windernte-2022-steigt-um-vier-prozent-gegenueber-dem-vorjahr-auf-gut-21
- Theuerkauff, D., Rivera-Ingraham, G. A., Lambert, S., Mercky, Y., Lejeune, M., Lignot, J.-H., & Sucré, E. (2020). Wastewater bioremediation by mangrove ecosystems impacts crab ecophysiology: In-situ caging experiment. *Aquatic Toxicology*, 218, 105358. https://doi.org/10.1016/j. aquatox.2019.105358
- Thoni, T., Beck, S., Borchers, M., Förster, J., Görl, K., Hahn, A., et al. (2020). Deployment of negative emissions technologies at the national level: A need for holistic feasibility assessments. *Frontiers in Climate*, 2. https://doi.org/10.3389/fclim.2020.590305
- Tittley, I. (1991). Seaweeds: Their environment, biogeography and ecophysiology, by Klaus Lüning, Wiley Interscience, 1991. xiii + 527 pp. Price: £70.50. Aquatic Conservation, 1(2), 189. https://doi.org/10.1002/aqc.3270010208
- UBA. (2023). Geoengineering-governance [text]. March 5, 2024, Retrieved from https://www.umweltbundesamt.de/en/topics/sustainabilitystrategies-international/environmental-law/international-environmental-law/geoengineering-governance
- UNFCCC. (2015). Paris agreement, report of the conference of the parties to the united nations framework convention on climate change. (21st Sess). 30 Nov.–13 Dec., Retrieved from https://unfccc.int/process-and-meetings/the-paris-agreement/the-paris-agreement

UNFCCC. (2016). The Paris Agreement. Retrieved from https://unfccc.int/documents/184656

Vesta. (2022). Retrieved February 23, 2024, from https://www.vesta.earth/



- Vielstädte, L., Linke, P., Schmidt, M., Sommer, S., Haeckel, M., Braack, M., & Wallmann, K. (2019). Footprint and detectability of a well leaking CO2 in the Central North Sea: Implications from a field experiment and numerical modelling. *International Journal of Greenhouse Gas Control*, 84, 190–203. https://doi.org/10.1016/j.ijgc.2019.03.012
- Wada, S., Omori, Y., Kayamyo, Y., Tashiro, Y., & Hama, T. (2015). Photoreactivity of dissolved organic matter from macroalgae. Regional Studies in Marine Science, 2, 12–18. https://doi.org/10.1016/j.rsma.2015.08.018
- Wang, W.-L., Fernández-Méndez, M., Elmer, F., Gao, G., Zhao, Y., Han, Y., et al. (2023). Ocean afforestation is a potentially effective way to remove carbon dioxide. *Nature Communications*, 14(1), 4339. https://doi.org/10.1038/s41467-023-39926-z
- Wayman, M., Hassan, K. E., & Al-Kuwari, M. S. (2021). Water footprint analysis of construction aggregates in Qatar. In *Towards a sustainable water future* (pp. 201–210). ICE Publishing. https://doi.org/10.1680/oicwe.65253.201
- Weatherley, N. S. (1988). Liming to mitigate acidification in freshwater ecosystems: A review of the biological consequences. *Water, Air, and Soil Pollution, 39*(3–4), 421–437. https://doi.org/10.1007/BF00279486
- Williamson, P. (2016). Emissions reduction: Scrutinize CO2 removal methods. Nature, 530(7589), 153–155. https://doi.org/10.1038/530153a
- Williamson, P., Wallace, D. W. R., Law, C. S., Boyd, P. W., Collos, Y., Croot, P., et al. (2012). Ocean fertilization for geoengineering: A review of effectiveness, environmental impacts and emerging governance. *Process Safety and Environmental Protection*, 90(6), 475–488. https://doi.org/ 10.1016/j.psep.2012.10.007
- Wu, J., Keller, D. P., & Oschlies, A. (2023). Carbon dioxide removal via macroalgae open-ocean mariculture and sinking: An earth system modeling study. *Earth System Dynamics*, 14(1), 185–221. https://doi.org/10.5194/esd-14-185-2023
- Wurgaft, E., Wang, Z. A., Churchill, J. H., Dellapenna, T., Song, S., Du, J., et al. (2021). Particle triggered reactions as an important mechanism of alkalinity and inorganic carbon removal in river plumes. *Geophysical Research Letters*, 48(11), e2021GL093178. https://doi.org/10.1029/ 2021GL093178
- Wüstenhagen, R., Wolsink, M., & Bürer, M. J. (2007). Social acceptance of renewable energy innovation: An introduction to the concept. *Energy Policy*, 35(5), 2683–2691. https://doi.org/10.1016/j.enpol.2006.12.001
- Yool, A., Shepherd, J. G., Bryden, H. L., & Oschlies, A. (2009). Low efficiency of nutrient translocation for enhancing oceanic uptake of carbon dioxide. *Journal of Geophysical Research*, 114(C8), 2008JC004792. https://doi.org/10.1029/2008JC004792
- Zimm, C., Mintz-Woo, K., Brutschin, E., Hanger-Kopp, S., Hoffmann, R., Kikstra, J. S., et al. (2024). Justice considerations in climate research. *Nature Climate Change*, 14(1), 22–30. https://doi.org/10.1038/s41558-023-01869-0

References From the Supporting Information

- Anderson, L., & Dyrssen, D. (1994). Alkalinity and total carbonate in the Arabian sea. Carbonate depletion in the red Sea and Persian Gulf. Marine Chemistry, 47(3), 195–202. https://doi.org/10.1016/0304-4203(94)90019-1
- Aspelund, A., & Jordal, K. (2007). Gas conditioning—The interface between CO2 capture and transport. International Journal of Greenhouse Gas Control, 1(3), 343–354. https://doi.org/10.1016/S1750-5836(07)00040-0
- Barbot, Y., Al-Ghaili, H., & Benz, R. (2016). A review on the valorization of macroalgal wastes for biomethane production. *Marine Drugs*, 14(6), 120. https://doi.org/10.3390/md14060120
- Beerling, D. J., Kantzas, E. P., Lomas, M. R., Wade, P., Eufrasio, R. M., Renforth, P., et al. (2020). Potential for large-scale CO2 removal via enhanced rock weathering with croplands. *Nature*, 583(7815), 242–248. https://doi.org/10.1038/s41586-020-2448-9
- Blomberg, A. E. A., Waarum, I.-K., Totland, C., & Eek, E. (2021). Marine monitoring for offshore geological carbon storage—A review of strategies, technologies and trends. *Geosciences*, 11(9), 383. https://doi.org/10.3390/geosciences11090383
- BMDV. (2016). Waterways as transport routes. November 20, 2023, Retrieved from https://bmdv.bund.de/SharedDocs/EN/Articles/WS/ waterways-as-transport-routes.html
- BMWK. (2023). Renewable energy sources in figures. February 13, 2024, Retrieved from https://www.bmwk.de/Redaktion/EN/Publikationen/ Energie/renewable-energy-sources-in-figures-2022.html
- Bruton, T., Lyons, H., Lerat, Y., Stanley, M., & Rasmussen, M. B. (2009). A review of the potential of marine algae as a source of biofuel in Ireland. *Sustainable Energy*. Retrieved from https://www.researchgate.net/publication/309185965_A_review_of_the_potential_of_Marine_Algae_as_a_Source_of_Biofuel_in_Ireland_Sustainable_Energy#fullTextFileContent
- BSH. (2009). Maritime spatial plan for the German EEZ in the North Sea. Retrieved March 5, 2024, Retrieved from https://maritime-spatial-planning.ec.europa.eu/practices/maritime-spatial-plan-german-eez-north-sea
- BSH. (2023). Flächenentwicklungsplan 2023 für die deutsche Nordsee und Ostsee. Retrieved from https://www.bsh.de/DE/THEMEN/Offshore/ Meeresfachplanung/Flaechenentwicklungsplan/_Anlagen/Downloads/FEP_2023_1/Flaechenentwicklungsplan_2023.pdf? __blob=publicationFile&v=1
- Buck, B. H. & Langan, R. (Eds.). (2017). Aquaculture perspective of multi-use sites in the open ocean. Springer International Publishing. https:// doi.org/10.1007/978-3-319-51159-7
- Carpenter, E. J. (1972). Nitrogen fixation by a blue-green epiphyte on pelagic sargassum. Science, 178(4066), 1207–1209. https://doi.org/10.1126/science.178.4066.1207
- Caserini, S., Pagano, D., Campo, F., Abbà, A., De Marco, S., Righi, D., et al. (2021). Potential of maritime transport for ocean liming and atmospheric CO2 removal. *Frontiers in Climate*, 3. https://doi.org/10.3389/fclim.2021.575900
- CESNI. (2020). Evaluating the energy requirement of inland vessels using energy efficiency indices. Retrieved from https://www.cesni.eu/wpcontent/uploads/2021/03/cesnipt_energyindex_en.pdf
- Chen, F., & Morosuk, T. (2021). Exergetic and economic evaluation of CO2 liquefaction processes. *Energies*, 14(21), 7174. https://doi.org/10. 3390/en14217174
- Chereau, E. (2019). Monitoring and evaluation of Sargassum collection operations summary report sargassum information hub. Retrieved from https://sargassumhub.org/monitoring-and-evaluation-of-sargassum-collection-operations-summary-report/
- Chowdhury, A., Naz, A., Bhattacharyya, S., & Sanyal, P. (2018). Cost–benefit analysis of 'Blue Carbon' sequestration by plantation of few key mangrove species at Sundarban Biosphere Reserve, India. *Carbon Management*, 9(6), 575–586. https://doi.org/10.1080/17583004.2018. 1518105
- Daniel-Gromke, J., Rensberg, N., Denysenko, V., Stinner, W., Schmalfuß, T., Scheftelowitz, M., et al. (2018). Current developments in production and utilization of biogas and biomethane in Germany. *Chemie Ingenieur Technik*, 90(1–2), 17–35. https://doi.org/10.1002/cite. 201700077
- Dickson, A. G., & Et, A. (2007). Guide to best practices for ocean CO2 measurement. https://doi.org/10.25607/OBP-1342



23284277, 2025, 4, Downloaded

- Doney, S. C. (2010). The growing human footprint on coastal and open-ocean biogeochemistry. Science, 328(5985), 1512–1516. https://doi.org/ 10.1126/science.1185198
- du Bois, P. B., Dumas, F., Voiseux, C., Morillon, M., Oms, P.-E., & Solier, L. (2020). Dissolved radiotracers and numerical modeling in NorthNorth European continental shelf dispersion studies (1982–2016): Databases, methods and applications. *Water*, 12(6), 1667. https://doi. org/10.3390/w12061667
- Du, F., Warsinger, D. M., Urmi, T. I., Thiel, G. P., Kumar, A., & Lienhard, J. H. V. (2018). Sodium hydroxide production from seawater desalination brine: Process design and energy efficiency. *Environmental Science and Technology*, 52(10), 5949–5958. https://doi.org/10.1021/ acs.est.8b01195
- Dutreuil, S., Bopp, L., & Tagliabue, A. (2009). Impact of enhanced vertical mixing on marine biogeochemistry: Lessons for geo-engineering and natural variability. *Biogeosciences*, 6(5), 901–912. https://doi.org/10.5194/bg-6-901-2009
- EBA. (2023). About biogas and biomethane. November 12, 2023, Retrieved from https://www.europeanbiogas.eu/about-biogas-and-biomethane/ Eiken, O., Davis, T. L., Landrø, M., & Wilson, M. (2019). Twenty years of monitoring CO2 injection at Sleipner. In *Geophysics and geo-sequestration* (pp. 209–234). Cambridge University Press. https://doi.org/10.1017/9781316480724.014
- Ejenstam, J. (2010). The lime industry, a potential business area for Kanthal. Retrieved from http://www.diva-portal.org/smash/record.jsf? pid=diva2%3A309947&dswid=-1900
- Fasihi, M., Efimova, O., & Breyer, C. (2019). Techno-economic assessment of CO2 direct air capture plants. *Journal of Cleaner Production*, 224, 957–980. https://doi.org/10.1016/j.jclepro.2019.03.086
- Flaathen, T. K., Gislason, S. R., Oelkers, E. H., & Sveinbjörnsdóttir, Á. E. (2009). Chemical evolution of the Mt. Hekla, Iceland, groundwaters: A natural analogue for CO2 sequestration in basaltic rocks. *Applied Geochemistry*, 24(3), 463–474. https://doi.org/10.1016/j.apgeochem.2008. 12.031
- Flipkens, G., Blust, R., & Town, R. M. (2021). Deriving nickel (Ni(II)) and chromium (Cr(III)) based environmentally safe olivine guidelines for coastal enhanced silicate weathering. *Environmental Science and Technology*, 55(18), 12362–12371. https://doi.org/10.1021/ACS.EST. 1C02974/ASSET/IMAGES/LARGE/ES1C02974_0003.JPEG
- FNR. (2019). Bioenergie in Deutschland: Zahlen und Fakten Feste Brennstoffe, Biokraftstoffe and Biogas. Fachangentur Nachwachsende Rohstoffe.
- Fraunhofer. (2021). Levelized cost of electricity-renewable energy technologies. November 23, 2023, Retrieved from https://www.ise.fraunhofer. de/en/publications/studies/cost-of-electricity.html
- Furre, A.-K., Eiken, O., Alnes, H., Vevatne, J. N., & Kiær, A. F. (2017). 20 Years of monitoring CO2-injection at sleipner. Energy Procedia, 114, 3916–3926. https://doi.org/10.1016/j.egypro.2017.03.1523
- Gao, G., Burgess, J. G., Wu, M., Wang, S., & Gao, K. (2020). Using macroalgae as biofuel: Current opportunities and challenges. *Botanica Marina*, 63(4), 355–370. https://doi.org/10.1515/bot-2019-0065
- Garcia, H. E., Locarnini, R. A., Boyer, T. P., Antonov, J. I., Baranova, O. K., Zweng, M. M., et al. (2013). World Ocean Atlas 2013. Vol. 4: Dissolved Inorganic Nutrients (phosphate, nitrate, silicate). S. Levitus, Ed.; A. Mishonov, Technical Ed., 4(September), 25. https://doi.org/10. 7289/V5J67DWD

GESAMP. (2019). High level review of a wide range of proposed marine geoengineering techniques. *GESAMP Reports and Studies*(98), 144. Ghaffour, N., Bundschuh, J., Mahmoudi, H., & Goosen, M. F. A. (2015). Renewable energy-driven desalination technologies: A comprehensive

- review on challenges and potential applications of integrated systems. *Desalination*, 356, 94–114. https://doi.org/10.1016/j.desal.2014.10.024 Gordon, P. (2019). Dubai to power desalination with solar. Retrieved November 23, 2023, from https://www.smart-energy.com/renewableenergy/dubai-to-power-desalination-with-solar/
- García-Rodríguez, L., & Gómez-Camacho, C. (2001). Exergy analysis of the SOL-14 plant (Plataforma Solar de Almería, Spain). *Desalination*, 137(1), 251–258. https://doi.org/10.1016/S0011-9164(01)00226-0
- Gude, V. G. (2016). Desalination and sustainability--an appraisal and current perspective. Water Research, 89, 87–106. https://doi.org/10.1016/j. watres.2015.11.012
- Gunson, A. J. (2013). Quantifying, reducing and improving mine water use. https://doi.org/10.14288/1.0071942
- Harris, R. N., & Higgins, S. M. (2008). A permeability estimate in 56 Ma crust at ODP hole 642E, Vøring plateau Norwegian sea. Earth and Planetary Science Letters, 267(1–2), 378–385. https://doi.org/10.1016/j.epsl.2007.11.055
- Howard, K. L., & Menzies, R. J. (1969). Distribution and production of sargassum in the waters off the Carolina coast. *Botanica Marina*, 12(1-4). https://doi.org/10.1515/botm.1969.12.1-4.244
- Hughes, A. D., Kelly, M. S., Black, K. D., & Stanley, M. S. (2012). Biogas from macroalgae: Is it time to revisit the idea? *Biotechnology for Biofuels*, 5(1), 86, https://doi.org/10.1186/1754-6834-5-86
- ITA group. (2021). Milo's desalination. November 23, 2023, Retrieved from https://itagroup.gr/en/milo-s-desalination/
- Jakobsen, J., Roussanaly, S., & Anantharaman, R. (2017). A techno-economic case study of CO2 capture, transport and storage chain from a cement plant in Norway. *Journal of Cleaner Production*, 144, 523–539. https://doi.org/10.1016/j.jclepro.2016.12.120
- Johnson, D. H., & Decicco, J. (1983). An artificial upwelling driven by salinity differences in the ocean. Retrieved from https://www.nrel.gov/ docs/legosti/old/2149.pdf
- Kaltschmitt, M., Hartmann, H., & Hofbauer, H. (Eds.). (2016). Energie aus Biomasse. Springer. https://doi.org/10.1007/978-3-662-47438-9 Kearns, D., Liu, H., & Consoli, C. (2021). Technology readiness and cost of CCS. Global CCS Institute. Retrieved from https://www.
- globalccsinstitute.com/wp-content/uploads/2021/03/Technology-Readiness-and-Costs-for-CCS-2021-1.pdf Keith, D. W., Holmes, G., St. Angelo, D., & Heidel, K. (2018). A process for capturing CO2 from the atmosphere. *Joule*, 2(8), 1573–1594. https://
- doi.org/10.1016/j.joule.2018.05.006
 Kellenberger, D., Althaus, H. J., Jungbluth, N., Künniger, T., Lehmann, M., & Thalmann, P. (2007). Life cycle inventories of building products.
 Final Report Ecoinvent Data v2. 0 No, 7. Retrieved from https://www.academia.edu/download/59905670/07_BuildingProducts20190701-59827-1ctcl96.pdf
- Kerrison, P. D., Stanley, M. S., Edwards, M. D., Black, K. D., & Hughes, A. D. (2015). The cultivation of European kelp for bioenergy: Site and species selection. *Biomass and Bioenergy*, 80, 229–242. https://doi.org/10.1016/j.biombioe.2015.04.035
- Kroeker, K. J., Kordas, R. L., Crim, R., Hendriks, I. E., Ramajo, L., Singh, G. S., et al. (2013). Impacts of ocean acidification on marine organisms: Quantifying sensitivities and interaction with warming. *Global Change Biology*, 19(6), 1884–1896. https://doi.org/10.1111/gcb.12179
- Kumar, A., Phillips, K. R., Cai, J., Schröder, U., & Lienhard, J. H. V. (2019). Integrated valorization of desalination brine through NaOH recovery: Opportunities and challenges. *Angewandte Chemie*, 131(20), 6570–6579. https://doi.org/10.1002/ange.201810469
- Kämpf, J., & Chapman, P. (2016). The functioning of coastal upwelling systems. In J. Kämpf & P. Chapman (Eds.), Upwelling systems of the world (pp. 31–65). Springer International Publishing. https://doi.org/10.1007/978-3-319-42524-5_2

- Lapointe, B. E., Brewton, R. A., Herren, L. W., Wang, M., Hu, C., McGillicuddy, D. J., et al. (2021). Nutrient content and stoichiometry of pelagic Sargassum reflects increasing nitrogen availability in the Atlantic Basin. *Nature Communications*, 12(1), 3060. https://doi.org/10.1038/s41467-021-23135-7
- Leedham, E. C., Hughes, C., Keng, F. S. L., Phang, S.-M., Malin, G., & Sturges, W. T. (2013). Emission of atmospherically significant halocarbons by naturally occurring and farmed tropical macroalgae. *Biogeosciences*, 10(6), 3615–3633. https://doi.org/10.5194/bg-10-3615-2013 Lin, R., Deng, C., Ding, L., Bose, A., & Murphy, J. D. (2019). Improving gaseous biofuel production from seaweed Saccharina latissima: The
- effect of hydrothermal pretreatment on energy efficiency. *Energy Conversion and Management*, *196*, 1385–1394. https://doi.org/10.1016/j. enconman.2019.06.044 Macovei, V. A., Voynova, Y. G., Becker, M., Triest, J., & Petersen, W. (2021a). Long-term intercomparison of two pCO2 instruments based on
- Macovei, V. A., Voynova, Y. G., Becker, M., Triest, J., & Petersen, W. (2021a). Long-term intercomparison of two pCO2 instruments based on ship-of-opportunity measurements in a dynamic shelf sea environment. *Limnology and Oceanography: Methods*, 19(1), 37–50. https://doi.org/ 10.1002/lom3.10403
- Malischek, R., & McCulloch, S. (2021). The world has vast capacity to store CO2: Net zero means we'll need it analysis. Retrieved March 5, 2024, from https://www.iea.org/commentaries/the-world-has-vast-capacity-to-store-co2-net-zero-means-we-ll-need-it
- Marinho, G. S., Holdt, S. L., Birkeland, M. J., & Angelidaki, I. (2015). Commercial cultivation and bioremediation potential of sugar kelp, Saccharina latissima, in Danish waters. *Journal of Applied Phycology*, 27(5), 1963–1973. https://doi.org/10.1007/s10811-014-0519-8
- Matter, J. M., Broecker, W. S., Stute, M., Gislason, S. R., Oelkers, E. H., Stefánsson, A., et al. (2009). Permanent carbon dioxide storage into basalt: The CarbFix pilot project, Iceland. *Energy Procedia*, 1(1), 3641–3646. https://doi.org/10.1016/j.egypro.2009.02.160
- McGrail, B. P., Spane, F. A., Amonette, J. E., Thompson, C. R., & Brown, C. F. (2014). Injection and monitoring at the Wallula basalt pilot project. *Energy Procedia*, 63, 2939–2948. https://doi.org/10.1016/j.egypro.2014.11.316
- Mirzabaev, A., Stringer, L. C., Benjaminsen, T. A., Gonzalez, P., Harris, R., Jafari, M., et al. (2022). Deserts, semiarid areas and desertification. In Climate change 2022 – impacts, adaptation and vulnerability: Working group II contribution to the sixth assessment report of the intergovernmental Panel on climate change (pp. 2195–2231). Cambridge University Press. Retrieved from. https://doi.org/10.1017/ 9781009325844.020
- Missimer, T. M., & Maliva, R. G. (2018). Environmental issues in seawater reverse osmosis desalination: Intakes and outfalls. *Desalination*, 434, 198–215. https://doi.org/10.1016/j.desal.2017.07.012
- Montserrat, F., Renforth, P., Hartmann, J., Leermakers, M., Knops, P., & Meysman, F. J. R. (2017). Olivine dissolution in seawater: Implications for CO 2 sequestration through enhanced weathering in coastal environments. *Environmental Science and Technology*, 51(7), 3960–3972. https://doi.org/10.1021/acs.est.6b05942
- Nagare, H., Fujiwara, T., Inoue, T., Akao, S., Inoue, K., Maeda, M., et al. (2012). Nutrient recovery from biomass cultivated as catch crop for removing accumulated fertilizer in farm soil. *Water Science and Technology*, *66*(5), 1110–1116. https://doi.org/10.2166/wst.2012.291
- Net Zero 2050 Team. (2021). Netto-Null-2050 web-Atlas. March 5, 2024, Retrieved from https://atlas.netto-null.org/ Nilsson, H., Van Overloop, J., Mehdi, R. A., & Pålsson, J. (2018). *Transnational maritime spatial planning in the North Sea: The shipping context*.
- Interreg North Sea Programme. Retrieved from https://northsearegion.eu/media/4836/northsea_finalshippingreport.pdf NPD. (2019). Carbon storage. March 5, 2024, Retrieved from https://www.sodir.no/en/facts/carbon-storage/
- Øi, L. E., Eldrup, N., Adhikari, U., Bentsen, M. H., Badalge, J. L., & Yang, S. (2016). Simulation and cost comparison of CO2 liquefaction. *Energy Procedia*, 86, 500–510. https://doi.org/10.1016/j.egypro.2016.01.051
- Paleologos, E. K., Al Nahyan, M. T., & Farouk, S. (2018). Risks and threats of desalination in the Arabian Gulf. IOP Conference Series: Earth and Environmental Science, 191(1), 012008. https://doi.org/10.1088/1755-1315/191/1/012008
- Palter, J., Cross, J., Long, M., Rafter, P., & Reimers, C. (2023). The science we need to assess marine carbon dioxide removal. EOS, 104. https:// doi.org/10.1029/2023EO230214
- Paris, B., Vandorou, F., Balafoutis, A. T., Vaiopoulos, K., Kyriakarakos, G., Manolakos, D., & Papadakis, G. (2022). Energy use in greenhouses in the EU: A review recommending energy efficiency measures and renewable energy sources adoption. *Applied Sciences*, 12(10), 5150. https:// doi.org/10.3390/app12105150
- Parkhurst, D. L., & Appelo, C. (1999). User's guide to PHREEQC (Version 2): A computer program for speciation, batch-reaction, onedimensional transport, and inverse geochemical calculations. *Water-Resources Investigations Report*, 99(4259), 312.
- Polo, G. (2012). On maritime transport costs, evolution, and forecast. Ciencia y Tecnología de Buques, 5(10), 19–31. https://doi.org/10.25043/ 19098642.57
- Qasim, M., Badrelzaman, M., Darwish, N. N., Darwish, N. A., & Hilal, N. (2019). Reverse osmosis desalination: A state-of-the-art review. *Desalination*, 459, 59–104. https://doi.org/10.1016/j.desal.2019.02.008
- Roussanaly, S., & Anantharaman, R. (2017). Cost-optimal CO 2 capture ratio for membrane-based capture from different CO 2 sources. *Chemical Engineering Journal*, 327, 618–628. https://doi.org/10.1016/j.cej.2017.06.082
- Raven, J. A. (2017). The possible roles of algae in restricting the increase in atmospheric CO 2 and global temperature. European Journal of Phycology, 52(4), 506–522. https://doi.org/10.1080/09670262.2017.1362593
- Renforth, P., Jenkins, B. G., & Kruger, T. (2013). Engineering challenges of ocean liming. *Energy*, 60(2000), 442–452. https://doi.org/10.1016/j.energy.2013.08.006
- Reyes-Lúa, A., Røe, I. T., & Jordal, K. (2021). CO₂ ship transport: Benefits for early movers and aspects to consider. Retrieved from https:// ccuszen.eu/sites/default/files/TG3_Briefing-CO2-ship-transport-Benefits-for-early-movers-and-aspects-to-consider.pdf
- Snæbjörnsdóttir, S. Ó., Sigfússon, B., Marieni, C., Goldberg, D., Gislason, S. R., & Oelkers, E. H. (2020). Carbon dioxide storage through mineral carbonation. Nature Reviews Earth and Environment, 1(2), 90–102. https://doi.org/10.1038/s43017-019-0011-8
- Schorcht, F., Kourti, I., Scalet, B. M., Roudier, S., & Sancho, L. D. (2013). Best available techniques (BAT) reference document for the production of cement, lime and magnesium oxide (p. 506). European Commission Joint Research Centre Institute for Prospective Technological Studies. Retrieved from https://core.ac.uk/download/pdf/38626619.pdf
- Schubert, P., Wein, J., & Bartsch, I. (2016). Laminaria bei Helgoland. Retrieved from https://www.schleswig-holstein.de/DE/fachinhalte/K/ kuestengewaesser/Downloads/vortrag16.pdf?_blob=publicationFile&v=1
- Smale, D. A., Pessarrodona, A., King, N., Burrows, M. T., Yunnie, A., Vance, T., & Moore, P. (2020). Environmental factors influencing primary productivity of the forest-forming kelp Laminaria hyperborea in the northeast Atlantic. *Scientific Reports*, 10(1), 12161. https://doi.org/10. 1038/s41598-020-69238-x
- Smith, P., Davis, S. J., Creutzig, F., Fuss, S., Minx, J., Gabrielle, B., et al. (2016). Biophysical and economic limits to negative CO2 emissions. *Nature Climate Change*, 6(1), 42–50. https://doi.org/10.1038/nclimate2870
- Socolow, R., Desmond, M., Aines, R., Blackstock, J., Bolland, O., Kaarsberg, T., et al. (2011). Direct air capture of CO2 with chemicals: A technology assessment for the APS Panel on public affairs. *American Physical Society*.



23284277, 2025, 4, Downloaded

- Tarantola, F., & Gentile, E. (2021). Analysis of slaked lime spreading methodologies for ocean alkalinity enhancement. *Civil and Environmental Engineering. Politecnico di Milano*. Retrieved from https://www.politesi.polimi.it/handle/10589/187644
- Techsci Research. (2023). UAE desalination plant market share, trends and growth report 2028F. Retrieved from https://www.techsciresearch. com/report/uae-desalination-plant-market/14913.html
- Tegtmeier, S., Krüger, K., Quack, B., Atlas, E. L., Pisso, I., Stohl, A., & Yang, X. (2012). Emission and transport of bromocarbons: From the west Pacific ocean into the stratosphere. *Atmospheric Chemistry and Physics*, *12*(22), 10633–10648. https://doi.org/10.5194/acp-12-10633-2012 Tosun, A., & Konak, G. (2015). Development of a model estimating energy consumption values of primary and secondary crushers. *Arabian*
- Journal of Geosciences, 8(2), 1133–1144. https://doi.org/10.1007/s12517-013-1260-3 UAE embassy. (2023). UAE energy diversification. Retrieved November 24, 2023, Retrieved from https://www.uae-embassy.org/discover-uae/
- climate-and-energy/uae-energy-diversification
- UAE Energy Report. (2015). The UAE state of energy report (p. 262). Ministry of Energy. Retrieved from https://play.google.com/store/books/ details?id=h5EwzgEACAAJ
- US Department of Energy. (2013). Limestone and crushed rock. In *ITP mining: Energy and environmental profile of the U.S. Mining industry* (pp. 9.1–9.12).
- Van Zanten, B. T., Brander, L. M., Gutierrez Torres, D., Uyttendaele, G. Y. P., Herrera Garcia, L. D., Patrama, D., & Kaczan, D. J. (2021). The economics of large-scale mangrove conservation and restoration in Indonesia. Retrieved from https://openknowledge.worldbank.org/ bitstream/handle/10986/37605/156489-brief.pdf?sequence=4
- Vangkilde-Pedersen, T., Anthonsen, K. L., Smith, N., Kirk, K., Neele, F., Van Der Meer, B., et al. (2009). Assessing European capacity for geological storage of carbon dioxide–the EU GeoCapacity project. *Energy Procedia*, 1(1), 2663–2670. https://doi.org/10.1016/j.egypro.2009. 02.034
- Viebahn, P., Scholz, A., & Zelt, O. (2019). The potential role of direct air capture in the German energy research program—Results of a multidimensional analysis. *Energies*, 12(18), 3443. https://doi.org/10.3390/en12183443
- Wang, M., Hu, C., Cannizzaro, J., English, D., Han, X., Naar, D., et al. (2018). Remote sensing of sargassum biomass, nutrients, and pigments. *Geophysical Research Letters*, 45(22). https://doi.org/10.1029/2018GL078858
- Weinberger, F., Paalme, T., & Wikström, S. A. (2020). Seaweed resources of the Baltic Sea, Kattegat and German and Danish North Sea coasts. *Botanica Marina*, 63(1), 61–72. https://doi.org/10.1515/bot-2019-0019
- Weiss, R. F. (1974). Carbon dioxide in water and seawater: The solubility of a non-ideal gas. *Marine Chemistry*, 2(3), 203–215. https://doi.org/10.1016/0304-4203(74)90015-2
- Wiese, F., Fridriksson, T., & Ármannsson, H. (2008). CO2 fixation by calcite in high-temperature geothermal systems in Iceland. Report from the Iceland Geosurvey (Ísor). Ísor-2008/003, Reykjavik. Retrieved from https://gogn.orkustofnun.is/Skyrslur/ISOR-2008/ISOR-2008-003.pdf
- Wollnik, R., Borchers, M., Seibert, R., Abel, S., Herrmann, P., Elsasser, P., et al. (2023). Dynamics of bio-based carbon dioxide removal in Germany (preprint). *Review*. https://doi.org/10.21203/rs.3.rs-3452150/v1
- Xu, G., Liang, F., Yang, Y., Hu, Y., Zhang, K., & Liu, W. (2014). An improved CO2 separation and purification system based on cryogenic separation and distillation theory. *Energies*, 7(5), 3484–3502. https://doi.org/10.3390/en7053484
- ZEP. (2019). The cost of subsurface storage of CO2. Retrieved from https://zeroemissionsplatform.eu/co2-storage-cost/